

Sami Järvinen ESSAYS ON PRICING COMMODITY DERIVATIVES

HELSINKI SCHOOL OF ECONOMICS

ACTA UNIVERSITATIS OECONOMICAE HELSINGIENSIS

A-238

Sami Järvinen ESSAYS ON PRICING COMMODITY DERIVATIVES

HELSINKI SCHOOL OF ECONOMICS

ACTA UNIVERSITATIS OECONOMICAE HELSINGIENSIS

A-238

© Sami Järvinen and Helsinki School of Economics

ISSN 1237-556X ISBN 951-791-861-5 ISBN 951-791-862-3 (Electronic dissertation)

> Helsinki School of Economics -HeSE print 2004

Acknowledgements

The writing of this dissertation has been a major undertaking. I have been fortunate to receive advice and support from many people during the process. I am grateful for all this help, and wish to thank the people involved.

There are three people who stand out in terms of their contribution to the finalization of this project. First, I would like to thank my advisor, Professor Jari Käppi. He was the person who introduced me to the doctoral program and gave valuable advice during the whole process. Second, I am grateful to my research colleague and business partner Harri Toivonen, with whom I share common research interests and whose ideas and support have been of great value. This dissertation also contains one of our joint papers. Third, I would like to thank my former colleague, Director Janne Saarikko. He has given the motivation for this dissertation from the commercial point of view and that has had a profound impact on the ideas and direction of this work. Taken together, the role of these three abovementioned people was crucial for this project to finally reach its conclusion. In addition, I would like to thank my pre-examiners, Professors Seppo Pynnönen and Kristian R. Miltersen, for insightful comments which significantly improved the final version of this thesis. Finally, Professor Vesa Puttonen had valuable suggestions on my dissertation and also provided much needed support at the final stages.

In addition, I have benefited from the many valuable comments of many people. At the early stages, I received theoretical help and support from Professor Esko Valkeila, who was always interested in discussing issues related to mathematical finance theory. Professor Staffan Ringbom gave insightful suggestions on the last essay of this dissertation when he held the role of a discussant in the Annual GSFFA Research Workshop. Finally, the participants of the GSFFA seminars in Helsinki and Nordic Symposium on Contingent Claims in Stockholm also provided valuable comments.

The research community in Helsinki School of Economics has been very supporting

and inspiring. Without it, this project would probably never had taken place. In particular, I would like to thank my fellow students, Sami Torstila, Markku Kaustia and Jussi Karhunen, for valuable discussions and support during the courses we had together. Antti Suhonen was very influential at the early stages and I benefited greatly from his ideas. Everybody at the Department of Finance and Accounting contributed by providing an inspiring working environment and collegial support. Professor Matti Keloharju deserves special acknowledgements for rooting high quality research standards as the core value of the department.

For financial support for this thesis, I would like to thank Helsinki School of Economics Foundation, Marcus Wallenberg's Foundation, the Research Foundation of OKO Bank, the Research Foundation of Säästöpankkiryhmä, the Paulo Foundation, the Finnish Foundation for Advancement of Securities Markets and Foundation for Economic Education.

Finally, I wish to thank my family for their patience, love and encouragement during this long process.

Helsinki, June 2004

Sami Järvinen

Contents

1.	Introduction	1
2.	Specific Features of Commodities	2
3.	Review of Literature on Term Structure Estimation	3
	3.1. Estimating the Term Structure of Interest Rates	3
	3.2. Estimating the Factor Structure of Forward Curve Movements	6
4.	Review of Literature on Pricing Commodity Contingent Claims	7
	4.1. Theory of Storage and Convenience Yield	7
	4.2. Summary of Pricing Models	.10
	4.3. Simple Approaches to Valuation	.12
5.	Summary of the Four Essays	.14
	References	17
F	PERAV 1.	
E E	CSSAY 1: Satimating the Forward Curve for Commodities	22
Е Е 1.	CSSAY 1: Satimating the Forward Curve for Commodities Preliminaries	22 .27
Е Е 1. 2.	CSSAY 1: Satimating the Forward Curve for Commodities Preliminaries Bootstrap Method	22 .27 .29
E E 1. 2. 3.	CSSAY 1: Setimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method	22 .27 .29 .36
E E 1. 2. 3. 4.	Sssay 1: Stimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method Optimization Method	22 .27 .29 .36 .39
E E 1. 2. 3. 4. 5.	Comparison of Methods.	22 .27 .29 .36 .39 .41
E E 1. 2. 3. 4. 5. 6.	CSSAY 1: Setimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method Optimization Method Comparison of Methods Empirical Results	22 .27 .29 .36 .39 .41 .45
E E 1. 2. 3. 4. 5. 6. 7.	CSSAY 1: Setimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method Optimization Method Comparison of Methods Empirical Results Extensions	 22 27 29 36 39 41 45 .52
E 1. 2. 3. 4. 5. 6. 7. 8.	Strimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method Optimization Method Comparison of Methods Empirical Results Extensions Conclusions	 22 27 29 36 39 41 45 52 53
E E 1. 2. 3. 4. 5. 6. 7. 8.	Strimating the Forward Curve for Commodities Preliminaries Bootstrap Method Spline Method Optimization Method Comparison of Methods Empirical Results Extensions Conclusions References	 22 .27 .29 .36 .39 .41 .45 .52 .53 .56

Es Pr Ev	SSAY 2: rincipal Components Analysis of Commodity Curves: vidence from Oil and Pulp Swap Markets	61
1. l	Model	64
2. I	Data	65
3. (Curve Estimation	68
4. I	Principal Components Analysis	70
5. I	Empirical Results	72
6. (Conclusions	80
1	References	82
Es Sir	SSAY 3: mple Binomial Model for Bermudan Swaptions	84
1. l	Notation, Swap Derivatives and Binomial Pricing	
	1.1. Standard European Swaption Formula	
	1.2. Binomial Approximation	90
2. 7	The Binomial Model For Bermudan Swaptions	92
:	2.1. Market Data	95
:	2.2. Variable Strike Bermudan	
-	2.3. Hedging Parameters	96
3. I	Numerical Examples	97
4. 4	Application to Commodity	
5. (Conclusion	104
I	References	106
Es Pr	SSAY 4: ricing European Commodity Swaptions	108
1. I	Basic Definitions	111
2. 7	The Pricing Formulas	114
3. I	Extensions	
4. I	Discussion	118
5. (Conclusions	120
I	References	121

1 Introduction

In recent years, commodity contingent claims have become increasingly popular and important hedging instruments in the financial markets characterized by varying degree of volatility. Previously, companies have primarily focused on hedging interest rate and foreign exchange risks. During recent years, hedging the commodity price risk of the production process has received more attention. As a result, the over-the-counter market for commodity contingent claims has grown remarkably. Therefore, the modelling of price processes and the development of new contingent claims pricing approaches are important area of research today.

This thesis aims to contribute to the literature on commodity contingent claims pricing through four interrelated essays. The first essay studies the estimation of the term structure of forward prices from the prices of traded long-term commodity swaps. The second essay investigates empirically the factor structure of the term structure movements using data from oil and pulp derivatives markets. Whereas previous studies have used futures prices, the data used in this study consists of swap market quotes. The forward curves used in the empirical study are derived using the method developed in the first essay. In the third essay, the focus is moved to the pricing of Bermudan swaptions using a simple binomial model. In addition to presenting an algorithm for pricing interest rate swaptions, an extension to handle commodity swaptions is also provided. Here again, the forward values of the variables are derived using the method developed in the first essay. Finally, the fourth essay is a joint work with Harri Toivonen. We concentrate on the necessary modification of the Black -76 formula in order to price European commodity swaptions. The central theme in all of the essays is the orientation towards using market data efficiently and the simplification of the modelling approach. Market oriented approaches towards modelling of the commodity price processes is a relatively new avenue. The vital ingredient to enable full utilization of those techniques

is extraction of the term structure of forward prices as the zero coupon term structure is an essential input in pricing interest derivative securities.

The structure of this introductory chapter is as follows: Section 2 presents the central concepts underlying the analysis of commodity contingent claims and describes the functioning of commodity markets in general. Section 3 reviews the existing literature on the term structure estimation. This earlier material is concerned almost exclusively with the estimation of the term structure of interest rates, the results of which form the essential basis for the results derived in this thesis. In Section 4, the objective is to review the existing literature on the theoretical models for pricing commodity contingent claims and to introduce the theory of storage. Finally, Section 5 presents a summary of the four essays in this thesis, with an emphasis on their contribution and extensions to the previous research.

2 Specific Features of Commodities

Commodities differ from financial assets in a profound way; unlike most financial assets, commodities are continuously produced and consumed. Moreover, from an economic point of view, commodities form an interesting asset class as production and consumption do not have to match each other in every period as commodities can be stored in the form of inventories. As an exception to this general rule some commodities, e.g. electricity and bandwidth, are not storable. Moreover, there are also other types of commodity classes such as agricultural and animal products that are perishable and hence in practice storable only for some period of time.

In equilibrium, supply increases if the price of commodity is high, and as a consequence higher cost producers enter the marketplace. Correspondingly, supply decreases if the price is low as some of the higher cost producers exit the marketplace. This economic activity is reflected by the well-known and documented phenomena of meanreverting commodity prices. Moreover, seasonal effects introduced by the nature of the production process are typical for many commodity markets. In particular, seasonal effects can be observed in agricultural commodity and electricity markets.

For the purpose of understanding futures prices, it is convenient to divide commodity futures contracts into the following two categories according to the underlying asset: investment and consumption commodities. Investment commodities, e.g. gold and silver, are held for investment purposes by a significant number of investors. Consumption commodities, e.g. oil, are held primarily for consumption purposes. In the case of consumption commodities, it is not possible to obtain the futures price as a function of the spot price and other observable variables. Hence, a parameter known as convenience yield becomes important.

Most commodity markets also have a well-functioning and liquid futures markets. One question that is often raised is whether the future price of an asset is equal to the expected future spot price. When the futures price is below the expected spot price, the situation is known as normal backwardation. Strong backwardation is the situation in which the futures price is below the current spot price. The situation where the futures price is above the expected future spot price is known as contango.

3 Review of Literature on Term Structure Estimation

3.1 Estimating the Term Structure of Interest Rates

The theoretical models for term structure estimation have been developed especially for interest rate markets where the amount and quality of data available has been good enough to facilitate the empirical comparison of the different methods. The first serious estimation method was proposed by McCulloch (1971). He developed a method for estimating the discount function by fitting a cubic spline to the data of government Bills and Bonds. The method proposed by McCulloch (1971) is especially appealing from the perspective of estimation: since the model is linear in the discount function, it can be estimated using simple linear regression. McCulloch (1975) applies the spline method to the estimation of the tax-adjusted yield curve. Carleton and Cooper (1976) estimated the discount function of the US Government securities, but without imposing any functional form. Their method produces a set of discrete discount bond prices that minimize the pricing error of the bonds. The estimation was carried out using linear regression.

Since the first steps in the estimation of the term structure of interest rates were taken, the research intensified and many new methodologies and extensions to the existing methods were proposed. One line of research extended the spline methodology to other functional forms. An important account in this respect is Vasicek and Fong (1982). The authors modified the spline estimation method so that the resulting discount function is exponential. The exponential form, the authors claim, avoids some of the problems associated with using polynomial forms in discount function estimation. Vasicek and Fong (1982) do not provide comprehensive evidence on the performance of their method. Shea (1985) criticized the exponential spline method and proved that it essentially produces a similar type of unstable forward rate functions as does the ordinary polynomial spline. Chambers, Carleton, and Waldman (1984) use exponential splines of varying lengths to estimate the present value functions. They let the sample decide which polynomial length best suits the given estimation problem. Maximum likelihood method is used to find the appropriate parameterization. Chambers, Carleton, and Waldman (1984) postulate that their method confirms that the use of a constant parameterization does not yield good results over heterogeneous samples. Their method,

however, is rather awkward to use in practical situations.

Research on improving the spline method, especially to yield smooth forward rate functions, is extended by Fisher, Nychka, and Zervos (1995) and Käppi (1997). These authors propose smoothing splines that impose a penalty on the target function which aims to produce a smoother forward rate function. The forward rate function is important especially for traders of derivative contracts, since the values of the underlying instrument often need to be derived from the forward rates (or prices). Fisher, Nychka, and Zervos (1995) claim, based on empirical comparisons against McCulloch (1971), that their method generally produces the most reliable forward curves and less biased results. The smoothing splines need to be estimated numerically.

Even though spline methods have received most of the attention, other methods for estimating the term structure of interest rates have also been proposed. One of the best known of these other approaches is Nelson and Siegel (1987). The proposed model expresses the instantaneous forward rates as a solution to the difference equation describing the movements of the spot rates. In the continuous time framework, the yields are just averages of these instantaneous forward rates, and the parameters of the solution equation can be estimated from, for example the yields of the government securities. Nelson and Siegel (1987) provide illustrations of their method using data on Treasury Bills. Depending on the chosen parameterization, there is a trade-off between fitting the front-end and back-end of the curve. Finally, the term structure of interest rates can be estimated using the so-called bootstrap method. This method is especially popular among traders of the derivatives contracts in the swap market, since par swap quotes provide ideal data to effectively bootstrap the term structure. As the quotes are always equally spaced, they do not move with the passage of time. Fama and Bliss (1987) is the best known academic treatment of the bootstrap methodology. They present two methods for estimating the discount functions: 1) Unsmoothed and 2) Smoothed. The former produces a piecewise linear discount function and prices the bonds used for estimation exactly, whereas the latter first estimates the unsmoothed discount factors as in the former case, but then tries to smooth out the discount function by fitting an approximating function through the discount factors.

Bliss (1996) conduct an extensive empirical comparison between the various methods for term structure estimation. The methods he tested were: 1) Unsmoothed Fama-Bliss, 2) McCulloch spline, 3) Fisher-Nychka-Zervos smoothing spline, 4) Extended Nelson-Siegel and 5) Smoothed Fama-Bliss. The most important conclusions of his study are that although the Unsmoothed Fama-Bliss method was able to fit the long term structures best, the method suffered from over-parameterization. Of the other methods, the Fisher-Nychka-Zervos method somewhat surprisingly performed the worst. The other methods were approximately equal.

3.2 Estimating the Factor Structure of Forward Curve Movements

Research on the commodity forward curve dynamics has used data almost exclusively on short term, exchange traded, contracts. Of these, among the first is Cortazar and Schwartz (1994), who studied the dynamics of the futures price curve of copper futures. They found that the factor structure of the copper futures curve was surprisingly similar to the factor structure of yield curve movements. Moreover, the explanatory power of the first two principal components was 97 percent. By contrast, Litterman and Scheinkman (1988) found that the explanatory power of first two principal components of the government yield curve movements was remarkably lower, around 90 percent. Clewlow and Strickland (2000) studied the factor structure of NYMEX oil futures and they found that three factors explained over 98 percent of the variation of the futures price movements in the period from 1998 to 2000. A recent paper by Tolmasky and Hindanov (2002) investigated the dynamics of the petroleum futures contracts. They found that especially for heating oil, seasonality is an important variable driving the factor structure, however the statistical significance is somewhat unclear. Crude oil and petroleum markets were not found to be affected by seasonality. Koekebakker and Ollmar (2001) studied the forward curve dynamics using data from the Nord Pool electricity derivatives exchange, using fitted forward curves. The explanatory powers they report are fairly low in comparison with other studies. They argue that the most likely reason for low explanatory power is the extremely complex dynamics of the electricity prices.

4 Review of Literature on Pricing Commodity Contingent Claims

4.1 Theory of Storage and Convenience Yield

Two different models have been introduced as alternative perspectives for the price formation of commodity futures prices. Working (1949) and Kaldor (1939) developed the theory of storage, which explains the difference between contemporaneous spot and futures prices on the basis of interest foregone in storing commodity product, warehousing costs and a convenience yield on inventory. The alternative model views commodity futures price as a combination of an expected risk premium and a forecast of the future spot price, see Cootner (1960). It has been argued that if hedgers tend to hold short positions and speculators tend to hold long positions, the futures price will be below the expected future spot price. This is because speculators require compensation for the risks they are bearing. They will only trade if there is an expectation that the futures price will rise over time. Hedgers, on the other hand, are prepared to enter into contracts where the expected payoff is slightly negative because they primarily aim at reducing the risk. Hence, if hedgers tend to hold long positions while speculators hold short positions, the futures price must be above the expected future spot price. To compensate speculators for the risk they are bearing, there must be an expectation that the futures prices will decline over time. Many studies on agricultural, wood, animal products and precious metals markets have shown results in favor of the former explanation, i.e. the theory of storage is able to explain the difference between the futures price and the contemporaneous spot price. This relation can be summarized into the following equation.

$$F(t,T) = (S(t) + W(t,T))e^{r(T-t)}$$

Where F(t, T) is the futures price at time t, for delivery of a commodity at time T. S(t)is the spot price at time t and W(t, T) is the present value of all the storage costs that will be incurred during the life of a futures contract. If the storage costs per unit are proportional to the price of the commodity, w(T, t), they can be regarded as providing a negative dividend yield

$$F(t,T) = S(t)e^{(r+w)(T-t)}$$

The theory of storage predicts that the return form purchasing the commodity at time tand selling it for delivery at time T, F(t,T) - S(t) equals the interest foregone, plus the marginal storage cost W(t,T), less the marginal convenience yield from an additional unit of inventory, $\delta(t,T)$. Or, equivalently

$$F(t,T) = S(t)e^{(r+w-\delta)(T-t)}$$

Where F(t,T) - S(t), or (F(t,T) - S(t))/S(t) is commonly known as basis, and r + w, or equivalently for consumption commodities r + w - y is referred to as the cost of carry. Often the above equation is written as in the form of net convenience yield, defined as convenience yield minus storage costs.

$$F(t,T) = S(t)e^{(r-\delta)(T-t)}$$

Convenience yield, δ , of a commodity is defined as the flow of services which accrues to the owner of the physical inventory, i.e. spot commodity, but not to the owner of a contract for future delivery, Brennan (1958). For investment assets such as gold, the convenience yield must be zero otherwise there will be arbitrage opportunities. However, a consumption asset behaves like an investment asset that provides a return equal to the convenience yield. The marginal convenience yield arises due to the fact that inventory can have productive value. The most obvious reason for this is that the owner of the commodity is able to determine where the commodity will be stored and when to liquidate inventory. Moreover, considering time lost and the costs incurred in ordering and transporting a commodity from one location to another, marginal convenience yield reflects both the reduction in costs of acquiring inventory and the value of being able to profit from temporary local shortages of the commodity through ownership of a larger inventory. Profit may also arise either from local price variations or from the ability to maintain production process despite local shortages of raw materials.

The convenience yield reflects the markets expectation concerning the future availability of the commodity. Hence, the greater the possibility that shortages will occur during the life of the futures contract, the higher the convenience yield. If users of the commodity have high inventories, there is very little chance of shortages in the near future and the convenience yield tends to be low. On the other hand, low inventories tend to lead to high convenience yields. This is especially true on those commodity markets where there are agents in the market who do not buy commodities only for production process purposes. This speculative storage activity can smooth the price and supply in this commodity market as the speculative storers buy the commodity when it is cheap and abundant, thus increasing the price, and then release their stores onto the market when the commodity is expensive and in short supply, thus reducing price. The convenience services yielded by particular inventory naturally depend upon the identity of the agent holding it. However, in theory, competition between potential storers will ensure that in equilibrium the convenience yield, net of any direct costs of storage, the marginal unit of inventory will be balanced across all storers. Then, assuming that there exists a positive inventory of the commodity, the relation between spot and futures price of the commodity will reflect this marginal net convenience yield.

In the case of Commodities with constant marginal costs of storage, the estimated convenience yield function is large and positive at low inventory levels and small and negative at high inventory levels, i.e. the theory of storage posits that the net rate of convenience yield depends upon the level of inventories. The economic reasoning behind this argument is as follows: Convenience yield is high because inventories are low and storage firms will tend to have an incentive to increase their investment in inventories which, in turn, will tend to reduce the convenience yield. Hence, the theory of storage predicts a negative relation between convenience yields and levels of inventories. Positive correlation between changes in the spot price and changes in the convenience yield of the commodity is induced by the level of inventories. As the inventories decrease, the spot price should increase due to the fact that the commodity is scarce, and the convenience yield should increase as the futures price will not increase as much as the spot price.

4.2 Summary of Pricing Models

Modelling commodity prices for the purposes of valuing derivative securities can be categorized into two distinct approaches, namely 1) convenience yield models and 2) forward price models. While this thesis concentrates exclusively on the latter, the former approach has been more common in the academic literature. It has sometimes been named as the traditional approach. The ideological inspiration of the approach is derived from the theory of spot rate modelling, pioneered by Vasicek (1977) and Cox, Ingersoll, and Ross (1981). The forward price modelling is the more market-oriented approach, since the quantities traded in the market are prices and therefore, it is intuitively more straightforward to model the prices directly. Similar developments have already taken place in the study of interest rate modelling. So-called market models (see Miltersen, Sandmann, and Sondermann (1997), Brace, Gatarek, and Musiela (1997) for the earliest developments and Musiela and Rutkowski (1997) and Jamshidian (1997) for generalizations) have largely replaced the traditional models of Hull and White (1990) and Black, Derman, and Toy (1990).

In a seminal paper, Gibson and Schwartz (1990) develop a two-factor model in which the dynamics of the commodity prices are determined as a function of the stochastic spot price of the commodity, and the stochastic instantaneous convenience yield. Schwartz (1997) extends the previous model by introducing also the stochastic interest rates. Furthermore, he provides empirical evidence on three different models using copper, oil and gold data. Evidence suggests that for the commercial commodities(copper and oil), mean reversion is an important characteristic of price movements, whereas for gold it is not. Miltersen and Schwartz (1998) take this idea a step further and introduce a model for forward/future convenience yield. The approach is similar to the infamous Heath, Jarrow, and Morton (1992) approach for interest rates. Hilliard and Reis (1998) introduce jumps to the spot price process.

Whereas in interest rate derivatives research, the forward rate based models have been more popular objects of study than spot rate models among both academics and practitioners, in commodity derivatives research, little attention has been devoted to developing forward price based models. The few exceptions include Reisman (1991) and Cortazar and Schwartz (1994). Both model the forward price processes as multidimensional geometric Brownian motion. The processes have zero drift, since under the risk-adjusted measure, it can be shown that forward prices must be martingales. Using Ito's lemma, one can make a transformation from the defined forward price process to the forward cost of carry process and vice versa, Cortazar and Schwartz (1994). Even though the theoretic notion of convenience yield is appealing from an economic point of view, practitioners working in the commodity derivatives industry will certainly find modelling the forward prices directly more intuitive.

4.3 Simple Approaches to Valuation

Complex pricing models are required to handle exotic derivative structures, whereas if market data on the plain vanilla instruments is available with reasonable accuracy, it is often more convenient to use standard Black and Scholes (1973) and Black (1976) models instead. To handle interest rate swaptions, modifications to the Black (1976) model have been proposed by Neuberger (1990) and Smith (1991). Since then, these formulas have become the market standard for pricing plain vanilla interest rate swaptions. Moreover, the practitioners use model implied volatilities in order to calibrate more complex models to the market data and, therefore, the exotic products can be priced consistently with actively traded instruments. This is essential, since the plain vanilla instruments are used for hedging exposures arising from trading the exotic products. Initially, it seemed that the Black (1976) model was inconsistent when applied to pricing interest rate products, since it assumed constant discounting. However, the measure change techniques, see Geman, Karoui, and Rochet (1995), formed theoretical justification for these simple models. In addition, the research efforts turned away from the traditional short rate models to modelling the stochastic evolution of the underlying variables (Libor and Swap rates) instead.

Black and Scholes (1973) model was a breakthrough in derivatives research. The formulas, however, applied only to European options. Therefore, applicable numerical methods were needed in order to handle more general option types. Algorithms for solving the Black-Scholes partial differential equation do exist, and Monte Carlo methods can be used to simulate Black-Scholes model stochastic differential equation in order to calculate path-dependent option prices. One of the best known numerical methods for calculating Black-Scholes model prices is Cox, Ross, and Rubinstein (1979). The authors derive a binomial lattice algorithm that converges at the limit to the Black-Scholes continuous time model as steps are increased to infinity. The binomial algorithm provided not only option price, but also the replicating portfolio calculations. The binomial model of Cox, Ross, and Rubinstein (1979) (CRR) has become the most important pedagogical tool in discrete time finance. In addition, it is one of the most applied numerical methods in actual option price.

Since the publication of the article by Cox, Ross, and Rubinstein (1979) article was published, researchers have devoted a great deal of interest to improving and modifying the simple binomial model in order to handle more difficult option structures and different stochastic processes. Jarrow and Rudd (1983) parameterize the algorithm so that the probabilities for up and down movements in the tree are always 1/2. This is sometimes a more convenient parameterization for constructing algorithms, for example if the interest rate and volatility parameters are not constant throughout the tree. Hull and White (1988) improve the accuracy of the binomial method by applying a control variate technique. In addition, Hull and White (1993) extend the binomial method in order to handle path-dependent options. Boyle (1988) and Boyle, Evnine, and Gibbs (1989) extend the CRR binomial model to handle options dependent on several assets. These and many more refinements and modifications to the original CRR binomial model have been proposed to improve pricing efficiency, handle complex exotic option contracts and deal with different distributional assumptions. One of the most important extensions to the model is the so called "Implied Binomial Tree" method, proposed by Rubinstein (1994). With this algorithm, the binomial tree structure is made consistent with the prices of traded options. Therefore, market data is used efficiently and other, more complex option structures can be priced with maximum utilization of market information.

5 Summary of the Four Essays

The first essay in this thesis develops methods for deriving the term structure of forward prices, or the forward curve, from the price quotes of commodity swaps. The importance of developing methods for forward curve extraction derives from the fact that dealing in commodity swaps has increased dramatically over the last decade. Existing research on commodity contingent claims pricing has concentrated exclusively on using the data from the futures markets, since it is readily available in convenient form. By contrast, the data (broker quotes) on the swap market contracts only became available in the late 1990's, and only for the most liquid markets. The focus of this first essay is mainly on extending the methods for interest rate term structure estimation to the commodity world. Empirical data from the oil and pulp derivatives markets is utilized to illustrate and compare the methods. The literature on the interest rate term structure estimation is vast, one of the seminal works being by McCulloch (1971), who applied the cubic spline method to the derivation of the discount function.

The main conclusions of this first essay are that the bootstrap method produces a very saw-toothed forward curve, particularly in cases where the averaging frequency of the index values over the settlement periods is high, i.e. daily. An example of this would be oil derivatives market. For most practical purposes then, the bootstrap method is not a realistic method to use. The standard cubic spline produces a smooth curve, but often with very large pricing errors. Finally, the optimization of fit method produces a smooth curve with acceptable pricing errors, so this method, or modifications of it, provides traders a practical means for deriving the forward curves for commodities.

The second essay examines the factor structure of the commodity term structure of forward price changes. This essay is an empirical examination, based on the estimation method developed in the first essay. To the best knowledge of the author, this is the first study on the commodity forward curve factor structure using swap contract quotes. There exists earlier studies on the commodity futures curve factor structure. Cortazar and Schwartz (1994) is one of the first studies examining the changes in the copper futures curve, and their findings were largely similar to corresponding studies done using the term structure of interest rate data. Koekebakker and Ollmar (2001) have studied the Nord Pool exchange's electricity contracts using a forward curve estimation method that is similar in spirit to the method used in this paper.

The empirical results of the second essay lead to a conclusion that the dynamics of the term structure of forward price changes is more complex when the data on the long-term derivatives contracts are used for estimation. Moreover, the dynamics of the front end of the curve is largely dictated by how the proxy for the spot index value is chosen in the forward curve estimation phase. It seems that bias is caused by the use of infrequently published index value data; if possible, one should therefore use the shortest future's quote as a proxy for the spot index value instead. In general, the results were similar to those found in earlier studies. The explanatory power of the first three principal components of Principal Components Analysis (PCA) applied to forward curve movements is around 89 percent for oil data and 84 percent for pulp data.

The data used in the first and second essays consists of weekly NBSK Risi pulp swap quotes covering the period from June 1998 to October 2001 and of monthly oil swap quotes (European Brent) covering the period from February 1997 to February 2002. The data was obtained from Nordea Bank Finland. The interest rate data used in estimation consists of Libor rates and zero rates. The European interest rate data consists of EuroLibor quotes and zero rates until 1.1.1999 and from that date onwards of Eurobor quotes and zero rates. The interest rate data was retrieved from DataStream.

The third essay in this thesis concentrates on the pricing of the derivatives contracts. In particular, a simple binomial method is presented for pricing Bermudan swaptions, an important subset of options, found especially in the interest rate markets. Existing algorithms and pricing methods are fairly complex to implement and calibrate to the prices of plain vanilla European Swaptions. This essay also extends the algorithm to be applicable in the commodity swaption markets. It is likely that in the near future, the Bermudan type swaptions will become interesting and popular hedging vehicles, as already is the case in the interest rate swaption market.

Finally, the fourth essay is a joint work with Harri Toivonen. We derive formulas for European commodity payer and receiver swaptions and provide extensions. The pricing formulas are direct modifications of the famous Black (1976) formula for options on the commodity futures. Similar formulas have been derived for the case of interest rate swaptions by Neuberger (1990).

References

- Black, F., 1976, "The Pricing of Commodity Contracts," Journal of Financial Economics, 3, 167–179.
- Black, F., E. Derman, and W. Toy, 1990, "A One-factor Model of Interest Rates and its Application to Treasury Bond Options," *Financial Analysts Journal*, 46, 33–39.
- Black, F., and M. Scholes, 1973, "The Pricing of Options and Corporate Liabilities," Journal of Political Economy, 81, 637–659.
- Bliss, R. R., 1996, "Testing Term Structure Estimation Methods," Working paper 96-12a, Federal Reserve Bank of Atlanta, November.
- Boyle, P., 1988, "A Lattice Framework for Option Pricing with Two State Variables," Journal of Financial and Quantitative Analysis, 23, 1–12.
- Boyle, P., J. Evnine, and S. Gibbs, 1989, "Numerical Evaluation of Multivariate Contingent Claims," *Review of Financial Studies*, 2, 241–250.
- Brace, A., M. Gatarek, and M. Musiela, 1997, "The Market Model of Interest Rate Dynamics," *Mathematical Finance*, 7, 127–155.
- Brennan, M. J., 1958, "The Supply of Storage," American Economic Review, 48, 50–72.
- Carleton, W. T., and I. A. Cooper, 1976, "Estimation and Uses of the Term Structure of Interest Rates," *Journal of Finance*, 31, 1067–1083.
- Chambers, D. R., W. T. Carleton, and D. W. Waldman, 1984, "A New Approach to Estimation of the Term Structure of Interest Rates," *Journal of Financial and Quantitative Analysis*, 19, 233–252.

- Clewlow, L., and C. Strickland, 2000, Energy Derivatives: Pricing and Risk Management. Lacima Publications, London, England, 1 edn.
- Cootner, P. H., 1960, "Returns to Speculators: Telser vs. Keynes," Journal of Political Economy, 68, 396–404.
- Cortazar, G., and E. S. Schwartz, 1994, "The Valuation of Commodity Contingent Claims," *Journal of Derivatives*, 1, 27–39.
- Cox, J., S. Ross, and M. Rubinstein, 1979, "Option Pricing: A Simplified Approach," Journal of Financial Economics, 7, 229–263.
- Cox, J. C., J. E. Ingersoll, and S. A. Ross, 1981, "The Relation Between Forward Prices and Futures Prices," *Journal of Financial Economics*, 9, 321–346.
- Fama, E. F., and R. R. Bliss, 1987, "The Information in Long-Maturity Forward Rates," American Economic Review, 77, 680–692.
- Fisher, M., D. Nychka, and D. Zervos, 1995, "Fitting the Term Structure of Interest Rates with Smoothing Splines," Working paper 95-1, Federal Reserve Board, Finance and Economics Discussion Series, January.
- Geman, H., N. E. Karoui, and J. Rochet, 1995, "Changes of Numeraire, Changes of Probability Measure and Option Pricing," *Journal of Applied Probability*, 32, 443– 458.
- Gibson, R., and E. S. Schwartz, 1990, "Stochastic Convenience Yield and Pricing of Oil Contingent Claims," *Journal of Finance*, 45, 959–976.
- Heath, D., R. A. Jarrow, and A. J. Morton, 1992, "Bond Pricing and the Term Structure of Interest Rates: A New Methodology for Contingent Claims Valuation," *Econometrica*, 60, 77–105.

- Hilliard, J. E., and J. Reis, 1998, "Valuation of Commodity Futures and Options under Stochastic Convenience Yields, Interest Rates and Jump Diffusions in the Spot," *Journal of Financial and Quantitative Analysis*, 33, 33–59.
- Hull, J., and A. White, 1988, "The Use of Control Variate Technique in Option Pricing," Journal of Financial and Quantitative Analysis, 23, 237–251.
- ——, 1990, "Pricing Interest Rate Derivative Securities," Review of Financial Studies, 3, 573–592.
- , 1993, "Efficient Procedures for Valuing European and American Path-Dependent Options," *Journal of Derivatives*, 1, 21–31.
- Jamshidian, F., 1997, "LIBOR and Swap Market Models and Measures," Finance and Stochastics, 1, 293–330.
- Jarrow, R., and A. Rudd, 1983, Option Pricing. Richard D. Irwin, Inc., Illinois, US.
- Kaldor, N., 1939, "Speculation and Economic Stability," Review of Economic Studies, 7, 1–27.
- Käppi, J., 1997, "Essays on Fixed Income Derivatives," Doctoral dissertation, Helsinki School of Economics.
- Koekebakker, S., and F. Ollmar, 2001, "Forward Curve Dynamics in the Nordic Electricity Market," Unpublished manuscript, Agder University College.
- Litterman, R., and J. Scheinkman, 1988, "Common Factors Affecting Bond Returns," Journal of Fixed Income, 1, 54–61.
- McCulloch, J. H., 1971, "Measuring the Term Structure of Interest Rates," Journal of Business, 44, 19–31.

— , 1975, "The Tax-Adjusted Yield Curve," Journal of Finance, 30, 811–830.

- Miltersen, K., K. Sandmann, and D. Sondermann, 1997, "Closed-form Solutions for Term Structure Derivatives with Lognormal Interest Rates," *Journal of Finance*, 52, 409–430.
- Miltersen, K. R., and E. S. Schwartz, 1998, "Pricing of Options on Commodity Futures with Stochastic Term Structures of Convenience Yields and Interest Rates," *Journal* of Financial and Quantitative Analysis, 33, 33–59.
- Musiela, M., and M. Rutkowski, 1997, "Continuous-time term structure models: Forward measure approach," *Finance and Stochastics*, 1, 293–330.
- Nelson, C. R., and A. F. Siegel, 1987, "Parsimonious Modeling of Yield Curves," *Journal of Business*, 60, 473–489.
- Neuberger, A., 1990, "Pricing Swap Options Using The Forward Swap Market," Working paper, London Business School.
- Reisman, H., 1991, "Movements of the Term Structure of Commodity Futures and the Pricing of Commodity Claims," Unpublished manuscript, Haifa University, Israel.
- Rubinstein, M., 1994, "Implied Binomial Trees," Journal of Finance, 49, 771–818.
- Schwartz, E. S., 1997, "The Stochastic Behavior of Commodity Prices: Implications for Valuation and Hedging," *Journal of Finance*, 52, 923–973.
- Shea, G. S., 1985, "Interest Rate Term Structure Estimation with Exponential Splines: A Note," *Journal of Finance*, 40, 319–325.
- Smith, D. R., 1991, "A Simple Method for Pricing Interest Rate Swaptions," Financial Analysts Journal, 47, 72–76.

- Tolmasky, C., and D. Hindanov, 2002, "Principal Components Analysis for Correlated Curves and Seasonal Commodities: The Case of the Petroleum Market," *Journal of Futures Markets*, 22, 1019–1035.
- Vasicek, A. O., 1977, "An Equilibrium Characterization of the Term Structure," Journal of Financial Economics, 5, 177–188.
- Vasicek, A. O., and H. G. Fong, 1982, "Term Structure Modeling Using Exponential Splines," *Journal of Finance*, 37, 339–348.
- Working, H., 1949, "The Theory of the Price of Storage," American Economic Review, 39, 1254–1262.

ESSAY 1:

Estimating the Forward Curve for Commodities

Abstract

This paper develops methods for estimating the term structure of forward prices and the term structure of convenience yields for general storable commodity classes on which tradable derivative contracts exist. The methodologies presented here can be used to infer inputs for the pricing of commodity contingent claims such as options. Another important area of application is the extraction of long-term convenience yields and forward prices for empirical studies on the stochastic processes of these variables. The estimation procedures are tested empirically using swap price quotes from the oil and pulp markets. The forward curve is the fundamental input in pricing contingent claims on any underlying assets with an active market. To date, there has been very little research on the estimation of the forward curve for commodity contracts of longer maturities. By contrast, estimation of the term structure of interest rates from the prices of bonds has been an active area of research pioneered by McCulloch (1971) and later extended by many researchers. In this paper, the estimation of the forward curve for commodity contracts from prices of actively traded over-the-counter derivative contracts is investigated, both theoretically and empirically. The potential uses of the estimation methods include applying new pricing models, testing hypotheses of the term structure of forward prices and convenience yields, and projecting futures prices for capital budgeting purposes.

Various approaches have been presented for estimating the term structure of interest rates. The best known and actively quoted method, presented by McCulloch (1971), is based on estimating the discount function using the cubic spline specification. The method is amenable to the use of linear regression due to equations being linear in parameters. The spline method has been expanded and investigated by many authors, Vasicek and Fong (1982), Shea (1984) and Chambers, Carleton, and Waldman (1984), among others. Fisher, Nychka, and Zervos (1995) and Käppi (1997) have extended the spline method further by exploiting the smoothness penalty in the estimation of the term structure of forward rates. Yet another well known approach to estimation of the term structure of interest rates has been presented by Nelson and Siegel (1987). They devised a parsimonious parameterization of the yield curve that can be used to generate yield curves with many different shapes. In addition to the above mentioned methods that rely on fitting to the data, there is also the well-known bootstrap method¹, which is based on iteratively solving a system of equations. The bootstrap method has also been extended by many authors - see, for example Fama and Bliss (1987). In addition to studying some of these well known methods of forward curve fitting for commodity forward curve estimation, this paper proposes a non-parametric minimization of the sum

of squared pricing errors and the squared difference of two adjacent maturity forward prices for extracting the forward prices from the prices of swap contracts. This method has a close counterpart from the interest rate yield and forward curve estimation field: Delbaen and Lorimier (1992) propose a non-parametric optimization method where the squared difference between adjacent forward rates is minimized.

The growing importance of obtaining reliable forward curves for commodity contracts is due to the expanding over-the-counter market for long term commodity derivatives. At present, active swap markets can be found for oil and some metals. At the same time, the modelling of the stochastic processes tailored for pricing contingent claims are moving from the traditional spot convenience yield models towards more market oriented approaches.² This transition has already been observed in the interest rate markets. One of the best known new models developed for pricing commodity contingent claims is by Miltersen and Schwartz (1998). Their approach is based on the celebrated Heath, Jarrow, and Morton (1992) model for the stochastic behavior of the instantaneous forward rate curve. Miltersen and Schwartz (1998) derive a model where the starting points are the initial term structure of interest rates and commodity futures prices along with the spot price of the commodity. After obtaining the risk neutral processes for each stochastic quantity, it is possible to price other contingent claims such as futures options within the model framework. A similar model is also presented by Hilliard and Reis (1998). Their approach extends the Miltersen and Schwartz (1998) model by including a jump component to the stochastic process for the spot price. In addition to these models that are based on the term structure of convenience yields, Reisman (1991) and Cortazar and Schwartz (1994) have developed a model that directly specifies the evolution of the term structure of futures prices. Cortazar and Schwartz (1994) apply the principal components analysis to estimate the model using the futures price data from the copper market. Both types of term structure models, convenience yield and futures prices, can be applied to price longer maturity contracts, as soon as the initial curves are available.

The main contribution of this paper is to provide a theoretical framework and applicable methods for obtaining the forward price curve from the prices of swaps. Several methodologies are presented: Bootstrapped Extraction, non-parametric Optimization of Fit, Cubic Spline based method and Cubic Spline including seasonality factor. The forward curve can be put into a linear form that conveniently enables the use of regression methods for minimization of the sum of squared errors. The Bootstrap method provides output forward curves that are exceedingly saw-toothed in the presence of average price swaps. The non-parametric Optimization of Fit method presented in this paper is flexible to use, and similar in spirit to calibrating model prices of options to the market prices. The methods presented here are primarily designed for estimation of the forward price curves, but the equivalence between the forward price and the convenience yield facilitates translation between the two quantities. In addition to the theoretical developments, empirical evidence on the performance of various methods is also presented as well as graphical illustrations. In this paper, the fitting methods are compared using data from the pulp and oil swap markets.

This paper is primarily interested in the pricing efficiency of the methods presented. Another area in forward curve estimation that needs to be investigated in future studies is the economic realism of the chosen methods for extracting the forward curves for commodities. In fact, it is often quoted that commodity prices are mean-reverting, follow business cycles and / or exhibit seasonal patterns depending on the type of commodity. Of the methods analyzed in this paper, only the Cubic Spline method including sin-cos functions aims at producing seasonal behavior on the extracted forward prices. All other methods are designed primarily to produce minimal pricing errors, in the same way as is done in the interest rate market. Optimization and basic Cubic Spline methods trade-off some of the pricing effectiveness against the smoothness of the forward price curve.

Another aim of this paper is to study empirically the methodologies presented with

the data from oil and pulp markets. By definition, the Bootstrap method prices each of the swaps in the sample exactly, so the empirical section of this paper focuses only on comparing the fitting methods. Evidence suggests that among the fitting methods proposed, the Optimization of Fit provides the best alternative for extracting the forward curve from the market prices of swaps. Friedman Rank Statistic strongly rejects the null hypothesis of no difference between the models. It is also found that using internal knot points with the spline method significantly improves the overall performance of the spline method. However, no method is satisfactory for real trading situations at all times. Even the Optimization of Fit method has to be applied with caution, due to intolerably large pricing errors with some term structures. An applicable remedy to resolve the pricing error issue is to use weighting to assing importance to specific portions of the curve. All the methods presented here are based on linear interpolation between the observed and derived curve points: This provides efficiency for estimation and simplicity for the formulas. Future research is needed to investigate the effect of various interpolation techniques to the results.

The paper is organized into eight sections. Section 1 introduces the basic definitions and terminology of the analysis framework. Section 2 describes the Bootstrap methodology for forward curve estimation, with consideration of both forward price curve and convenience yield term structure specifications. Section 3 presents the Cubic Spline estimation method and Section 4 develops the Optimization of Fit method. The new methodologies are compared in Section 5 and in Section 6 the models are fitted to the data and empirical results reported. Section 7 discusses some possible extensions to the methodologies. Finally, section 8 concludes the paper.

1 Preliminaries

The aim of this paper is to derive theoretical commodity forward price curves from the prices of traded swap contracts. This chapter reviews the fundamental concepts of commodity swaps. The basic elements needed for the analysis are the par swap quotes G(t,T) for all maturities $T, t \leq T$, the spot price of the underlying commodity S_t , the forward prices of the commodity, F(t,T), for all maturities $T, t \leq T$. In addition, we set $F(t,t) = S_t$. Futures prices, f(t,T), and the pure discount bond prices P(t,T), for all maturities $T, t \leq T$, are also essential parts of the framework. P(t,T) denotes the present value, at time t of one unit of money received at time T. The analysis assumes constant interest rates throughout, so theoretical forward and futures prices are equal, as shown by Cox, Ingersoll, and Ross (1981). Rather than theoretical futures prices, in the current setup the futures prices, f(t,T), are taken to represent exchange traded futures contracts that can be utilized in the curve generation process.

The starting point of the analysis is the valuation formula for commodity swaps. Typically, the swap contracts quoted in the commodity derivatives markets are such that fixed payment streams are exchanged, in monthly or quarterly intervals, for floating price payments. The floating price payments are usually based on some index or spot value of the commodity, calculated and defined by a reliable market information provider. Additionally, sometimes the futures price quotes are used to determine the floating price settlements. The distinguishing feature of a typical commodity swap contract from a standard interest rate swap contract is the averaging of the floating prices in determination of the final settlement prices. In more liquid markets, such as the oil derivatives markets, the averaging frequency is usually daily, whereas in the less liquid markets the observations are typically made weekly or monthly. Single price observation swaps also exist, but they occur less frequently in the commodity derivatives markets.³

Consider a standard commodity swap contract, with maturity T_N and a fixed contract

price $G(t, T_N)$. The present value of this fixed leg of the swap, given the discount bond prices P(t, T) is given by

$$V_{fx}^{N} = G(t, T_{N}) \sum_{i=1}^{N} P(t, T_{i})$$
(1)

i.e, simply a sum of the present values of the payments. The other side of the swap contract, the floating leg, is composed of the present values of the yet unknown average prices of the commodity at settlement dates T_i , i = 1, 2, ...N and $T_1 < T_2 < ... < T_N$. In mathematical notation

$$V_{fl}^{N} = \sum_{i=1}^{N} F_{A}(t, T_{i}) P(t, T_{i})$$
(2)

where A is used to indicate that the swap contract has average- based floating price settlement. If we know the forward price curve, the par swap prices for all maturities, $T, t \leq T$, could be obtained by setting the values of the floating and fixed legs to be equal. Solving for the par swap price $G(t, T_N)$ gives

$$G(t, T_N) = \frac{\sum_{i=1}^{N} F_A(t, T_i) P(t, T_i)}{\sum_{i=1}^{N} P(t, T_i)}$$
(3)

which shows that the par swap price is defined by the ratio of the discounted value of the floating price payments and an annuity. Therefore, as with interest rate swaps, the par swap price can be interpreted as the present value weighted average of the forward prices.

The traditional approach of modelling the forward prices of commodities is based on defining the net convenience yield, which is defined as the net "income", accruing to the owner of the physical commodity, Brennan (1958). The relation between the spot price of commodity, S_t , and the forward price of maturity, T, is given by the well known formula

$$F(t,T) = S_t e^{(r(t,T) - \delta(t,T))(T-t)}$$
(4)

which can be written, using the familiar formula $e^{-r(t,T)(T-t)} = P(t,T)$, as follows:

$$F(t,T) = \frac{S_t e^{-\delta(t,T)(T-t)}}{P(t,T)}$$

$$\tag{5}$$

showing that there is a one-to-one correspondence between the net convenience yields and forward prices. Therefore, either the term structure of forward prices or the term structure of convenience yields can be estimated from the prices of swaps (and futures).

2 Bootstrap Method

In this section, the whole procedure of deriving the implied average forward prices and direct forward prices (forward prices without averaging) is presented. The procedure is based on direct application of the net present value rule in order to iteratively solve the system of equations. Assume that par swap quotes $G(t, T_i)$ for maturities $T_i = 1, 2, ...$ years are observed in the market. In addition, the spot price of the commodity, S_t , is observed. The procedure begins with interpolation of the par swap curve in order to find the par swap prices for maturities corresponding to the settlement dates of the swap payments. Interpolation of the swap curve is necessary because there are not enough swap quotes to cover all the settlement dates. For example, if the market convention is fixed price versus quarterly settled averages of daily price observations, then the par swap curve needs to be interpolated in order to find prices for three month, six month, nine month, etc par swaps. The first implied average forward price sets the initial seed for the Bootstrap method.

Having interpolated the par swap curve to find the swap price corresponding to the
first settlement date of the one year swap, we can use the standard net present value method to obtain the first implied average forward price. As at contract initiation, the value of a swap is zero, so it follows that

$$(G_A(t, T_1) - F_A(t, T_1))P(t, T_1) = 0$$

and solving for $F_A(t, T_1)$ gives

$$F_A(t, T_1) = G_A(t, T_1)$$
 (6)

Hence, the first implied average forward price is just the value interpolated from the par swap curve. Therefore, the procedure can now be continued in order to find the second point. For this one can use the net present value with the six month par swap, that is:

$$\left(G_{A}(t,T_{2})-F_{A}(t,T_{1})\right)P(t,T_{1})+\left(G_{A}(t,T_{2})-F_{A}(t,T_{2})\right)P(t,T_{2})=0$$

with the solution

$$F_A(t, T_2) = G_A(t, T_2) + \left(G_A(t, T_2) - F_A(t, T_1)\right) \frac{P(t, T_1)}{P(t, T_2)}$$
(7)

The full procedure, generalizing up to $T_n \leq T_N$ -maturity par swap quote, can be expressed with a simple formula

$$F_{A}(t,T_{n}) = G_{A}(t,T_{n}) + \sum_{i=1}^{n-1} \left(G_{A}(t,T_{n}) - F_{A}(t,T_{i}) \right) \frac{P(t,T_{i})}{P(t,T_{n})}$$
(8)

Using equation (8), the whole term structure of implied average forward prices can be recovered. Having obtained the implied average forward prices, the next step is to obtain the implied direct forward prices. This needs to be done because the term structure of implied average forward prices is not flexible enough to be used for pricing general commodity derivative products, since there may be different averaging periods or no averaging at all in other derivatives structures.⁴ Let k denote the number of observations for the arithmetic average on a single settlement period. (In practice, kis the number of bank days on a single settlement period, if the averaging frequency is daily, though other conventions are also possible.) The following shows the steps for finding the implied direct forward prices. First it should be noted that there is a problem of non-uniqueness of the direct forward prices, since our task is to produce $F_A(t, T_i)$ from a set of possible direct forward prices, $F(t, T_i^j)$. The average can therefore be arrived at by any suitable combination of $F(t, T_i^j)$'s. Therefore, in order to force unique direct forward prices from a set of average prices, impose the linear interpolation constraint as follows

$$F_A(t, T_i) = \frac{\sum_{j=1}^k F(t, T_i^j)}{k}$$

= $\frac{\sum_{j=1}^k \left((F(t, T_i) - F(t, T_{i-1})) \frac{j}{k} + F(t, T_{i-1}) \right)}{k}$

which can be rewritten

$$F_A(t, T_i) = \frac{F(t, T_i) \sum_{j=1}^k \frac{j}{k} + \left(k - \sum_{j=1}^k \frac{j}{k}\right) \times F(t, T_{i-1})}{k}$$

and using the fact that $\sum_{i=1}^{k} i = k(k+1)/2$, the equation can simplified further:

$$F_A(t,T_i) = \frac{1}{2} \left[\left(1 + \frac{1}{k} \right) F(t,T_i) + \left(1 - \frac{1}{k} \right) F(t,T_{i-1}) \right]$$
(9)

This can now be solved inductively, so that a full term structure of direct forward

prices can be extracted. The first step is to solve for $F(t,T_1)$. This can be done, because we know that the first direct forward price is equal to the spot price (which is directly observable), so $F(t,T_0) = S_t$. Finally, use equation (9) and $F_A(t,T_1)$, which was already solved in the first phase when the implied average forward prices were obtained. This process is continued iteratively until the whole term structure of direct forward prices is derived. Figures 1 and 2 show examples of forward price curves using the method presented above. The data is from the Brent oil market in both upward and downward sloping swap price curve environments. Also note that the observation frequency of the index values for the settlement prices of the floating legs can be daily, weekly, monthly etc. If the payment frequency is equal to the observation frequency, then clearly $F_A(t,T) = F(t,T)$, and there is no need to complete the last step of the Bootstrap method.

The main problem with the Bootstrap method is that the forward curve it produces is exceedingly saw-toothed. This is even more so if the underlying swap market data consists of average settlement contracts. The average forward prices are solved for first with the Bootstrap method, i.e. the first phase of the algorithm derives a forward curve for implied average prices. In order to force the direct forward curve to yield those same average prices, it must attain extreme values. An example clarifies this: Consider that the spot index value is 100 units. The settlement frequency is daily with 22 observations in the current one month period. The implied average forward price for the first settlement is 105 units. Then applying formula (9) yields the direct forward price of 109.57 units.

Additionally, the forward price curve can also be derived directly in terms of the net convenience yields. In this case, one starts with the implied average forward price curve that has already been derived at this point. The term structure of cost of carry can then

Figure 1: OIL SWAP AND FORWARD CURVES FOR 3/31/99, bootstrap forward prices method



Figure 2: OIL SWAP AND FORWARD CURVES FOR 3/31/02, bootstrap forward prices method



be obtained by solving iteratively

$$F_A(t, T_i) = \frac{S_t \sum_{j=1}^k e^{c(t, T_i)(\Delta j + T_{i-1})}}{k}$$
(10)

where S_t is the spot price of the commodity, $c(t, T_i)$ is the cost of carry parameter for the period of average forward. Δ divides the period into intervals, from the start to the end of averaging. For example, if the observations are taken daily, then $\Delta = 1/365$. kagain denotes the number of observations on a single settlement period from T_{i-1} to T_i . Equation (10) has to be solved iteratively for $c(t, T_i)$. That can be easily done using, for example the Newton-Raphson scheme. After the full term structure of cost of carry is obtained, various swap structures, including the standard swap products, can be priced.

Using the fact that $e^{c(t,T)} = e^{r(t,T)-\delta(t,T)} = e^{-\delta(t,T)}/P(t,T)$, together with (10), it is possible to derive the net convenience yield $\delta(t,T)$ term structure, using the formula given below

$$F_A(t, T_i) = \frac{S_t \sum_{j=1}^k \frac{e^{-\delta(t, T_i)(\Delta_j + T_{i-1})}}{P(t, T_{(i-1+\Delta_j)})}}{k}$$
(11)

Note that the net convenience yield parameter is expressed in terms of continuously compounded return. This is due to the fact that the theoretical models of the net convenience yield are defined this way. If one wants to use some other compounding style, the adjustments to the formulas are then straightforward. However, the price average is always defined in terms of the arithmetic average. Therefore, it must be in the form presented above.

Figures 3 and 4 show examples of forward curves derived using the bootstrapped net convenience yield method.

Figure 3: OIL SWAP AND FORWARD CURVES FOR 3/31/99, bootstrap convenience yield method



Figure 4: OIL SWAP AND FORWARD CURVES FOR 3/31/02, bootstrap convenience yield method



3 Spline Method

In this paper, the spline form used is the standard cubic functional, similar to that used by McCulloch (1971), though more general forms could be applied. Results are derived using general functional forms. It is demonstrated that the forward price term structure can be obtained using simple linear regression, analogously to discount function estimation. The net present value rule requires that the fixed leg and floating leg present values are equal, for each n = 1, 2, ..., N

$$G(t,T_n)\sum_{i=1}^{n} P(t,T_i) = \sum_{i=1}^{n} F_A(t,T_i)P(t,T_i)$$
(12)

In order to fit a spline function to the forward curve by linear regression, k continuously differentiable $g_j(T)$ functions must be defined. In the following, it is assumed that t = 0. Hence, the forward curve can be represented as a constant term and a linear combination of the functions $g_j(T)$

$$F_A(0,T) = \alpha_0 + \sum_{j=1}^k \alpha_j g_j(T)$$
(13)

Since the forward price for F(t, t) must equal S_t , it is therefore required that $\alpha_0 = S_t$. Combining equations (12) and (13), the function to be estimated becomes, for each n = 1, 2, ..., N

$$V_{fl,n} = \sum_{i=0}^{n} F_A(0, T_i) P(0, T_i)$$

=
$$\sum_{i=0}^{n} \sum_{j=0}^{k} \alpha_j g_j(T_i) P(0, T_i)$$

=
$$\sum_{j=0}^{k} \alpha_j G_{n+1,k+1}$$

which can be expressed in vector notation as follows

$$\mathbf{V}_{fl} = \mathbf{G}\mathbf{A} \tag{14}$$

where

$$\mathbf{G} = \begin{pmatrix} 1 & g_1(0) & g_2(0) & \cdots & g_k(0) \\ P(0,T_1) & P(0,T_1)g_1(T_1) & P(0,T_1)g_2(T_1) & \cdots & P(0,T_1)g_k(T_1) \\ \sum_{i=1}^2 P(0,T_i) & \sum_{i=1}^2 P(0,T_i)g_1(T_i) & \sum_{i=1}^2 P(0,T_i)g_2(T_i) & \cdots & \sum_{i=1}^2 P(0,T_i)g_k(T_i) \\ \cdots & \cdots & \cdots & \cdots \\ \sum_{i=1}^N P(0,T_i) & \sum_{i=1}^N P(0,T_i)g_1(T_i) & \sum_{i=1}^N P(0,T_i)g_2(T_i) & \cdots & \sum_{i=1}^N P(0,T_i)g_k(T_i) \end{pmatrix}$$

and **A** is a vector of coefficients. The fact P(t,t) = 1 has been used to simplify the notation. Figures 5 and 6 show examples of forward curves derived using the standard Cubic Spline method.

The spline method can be made more flexible in order to handle difficult curve shapes by introducing knot points. Therefore, the spline method will be extended to piecewise polynomial functions for the empirical part of this paper. In the empirical study, only two internal evenly spaced knot points are used. It is left for further research to investigate the effect of knot positioning and the optimal amount of knots. Details of the effects on the estimation of the resulting regression are given by Suits, Mason, and Chan (1978).

In addition to the standard piecewise cubic spline, the sine and cosine functions are also used in order to capture possible seasonality in the data. It is often postulated that commodity prices exhibit mean reversion. Therefore, that possibility will also be considered in this study. In order to implement seasonality with curve estimation, a combination of cubic polynomial and an expansion of two sine and cosine functions are used. Since with polynomial, we have k = 4, it follows that there will be four additional Figure 5: OIL SWAP AND FORWARD CURVES FOR 3/31/99, splining forward curve method



Figure 6: OIL SWAP AND FORWARD CURVES FOR 3/31/02, splining forward curve method



coefficients to be estimated

$$g_{k+1}(T) = \cos\left(\frac{2*T*\pi}{\beta}\right)$$
$$g_{k+2}(T) = \sin\left(\frac{2*T*\pi}{\beta}\right)$$
$$g_{k+3}(T) = \cos\left(\frac{4*T*\pi}{\beta}\right)$$
$$g_{k+4}(T) = \sin\left(\frac{4*T*\pi}{\beta}\right)$$

where β is the period factor. In principle, β can either be estimated from the historical data or used as one parameter in the fitting procedure.

As is the case with the bootstrap method, the direct forward prices F(t,T) can be obtained by imposing the linear interpolation constraint and then using (9). The direct forward price curve will still be a saw-toothed function, however. If one wants to get a smooth direct forward price curve, then the floating price leg value has to be transformed to be dependent directly on the direct forward prices (instead of the implied average forward prices). An obvious choice is to use equation (9), to impose linear interpolation constraint. The spline can then also be configured to handle this situation.

4 Optimization Method

This section develops the Optimization of Fit method for extracting forward curve information from the prices of commodity swaps. The method is based on the minimization of the sum of squared errors. The aim is to fit the observed swap prices as precisely as possible while maintaining a reasonable degree of smoothness in the output forward curve. The proposed target function of the minimization problem is

$$\min_{\{F_i\}_{i=1}^N} \left[\sum_{i=1}^N \left(V_{fl}(t, T_i) - V_{fx}(t, T_i) \right)^2 + \lambda \sum_{i=1}^N \left(F(t, T_i) - F(t, T_{i-1}) \right)^2 \right]$$
(15)

where the fundamental idea is to match the swap values from the forward curve to the observed swap values, and simultaneously controlling that consecutive forward prices are not too far apart. To motivate this choice, let us note that the desire of a trader is often to produce a realistic looking, or at least not excessively vibrating, forward curve with which to price other products. Similar ideas have been used in the interest rate market in the spline form, for example by Fisher, Nychka, and Zervos (1995) and Käppi (1997). Delbaen and Lorimier (1992) use similar methodology to estimate the forward rate curve from the prices of bonds. Their objective is to minimize the squared difference between two adjacent maturity forward rates. Additionally, they impose a constraint that the relative pricing errors lie within a given tolerance. The approach presented in this paper is closely related to the latter method. The empirical results show that the method works well in comparison with the spline method. This suggests that, especially in the commodity markets where rather extreme and awkward curve shapes are very common, it is beneficial to have a more flexible fitting method.

The parameter λ is introduced to adjust the weight put on the forward curve smoothness. Note that the formulation uses the direct forward prices, so the output curve from the optimization problem is immediately of desired form. Moreover, it follows from the linear interpolation constraint that the derivatives for the optimization problem are straightforward to calculate. Therefore, analytic derivatives can be used in the actual minimization - see Appendix for the calculation of the derivatives. The optimization problem can be solved using any suitable algorithm. The method utilized in this paper is based on a simple gradient search with constant step size and analytical derivatives. While not the fastest possible algorithm, the convergence properties are good and the global minimum is almost sure to be attained for any curve shapes. For an excellent description on the various optimization algorithms, see Press, Teukolsky, Vetterling, and Flannery (1992).

In choosing λ , two points need to be mentioned. First, if one picks $\lambda = 0$, then the Optimization of Fit method will converge to the bootstrapped curve and the target function value will be equal to zero. I.e. exact match can be obtained through the Optimization method. Alternatively, if a high enough λ is chosen, then, starting from the spot value S_t , the final forward curve will be flat. Hence, a reasonably shaped and correctly pricing forward curve will result only if one picks a reasonable value for λ . For the purposes of this study, λ is set equal to one. The decision is based on experimenting with the algorithm, not explicitly on any objective criteria. In further research, implementing some penalty function like in Fisher, Nychka, and Zervos (1995) might yield interesting results. Figures 7 and 8 show examples of forward curves obtained using the Optimization of Fit method.

5 Comparison of Methods

Four methods for estimating the forward curve for commodities are developed. The Bootstrap methods use the analogy from the term structure of interest rate estimation research to iteratively solve a system of equations to obtain the forward prices either directly, or indirectly through convenience yield. Also, the Cubic Spline method derives analogy from the term structure of interest rates estimation literature to specify equations for commodity forward curve estimation. The Optimization of Fit method, however, is more like a calibration procedure, where the forward curve is derived by minimizing the sum of squared pricing errors. Weighting can also be utilized to express the relative importance of each swap price and the forward curve smoothness. Figure 7: OIL SWAP AND FORWARD CURVES FOR 3/31/99, optimization of fit method



Figure 8: OIL SWAP AND FORWARD CURVES FOR 3/31/02, optimization of fit method



• Bootstrap Forward Curve Method

The Bootstrap forward curve method is a recursive method to solve for implied average forward prices and, further, direct forward prices. At each step, the next forward price is obtained from the swap pricing equation and previous forward prices. In order to find the swap prices for intermediate points, it is necessary to interpolate the swap prices since there are not enough quotes to span all periods. An easy choice is to interpolate linearly, but other choices are also possible. Moreover, it is also necessary to impose interpolation condition on the derived implied average prices, since an average forward price is composed of a set of direct forward prices. The Bootstrap forward curve method prices market swap quotes exactly, but the resulting curve is intolerably jagged.

• Bootstrap Convenience Yield Method

Bootstrapping the convenience yield term structure results in a forward price curve that behaves better than bootstrapping the forward prices directly. However, it is not as convenient to use in real applications, since the specification presented in this paper is such that the spot convenience yield is constant for each period. Therefore, if one interpolates the forward curve, as is customary in trading systems, the market swaps are no longer priced exactly. However, correct results can be obtained by appropriately constructing the algorithms used in this approach.

• Cubic Spline Method

There are many possibilities to specify splining of the forward price curve in order to find the best suited result for each particular need. One can put the spline on the implied average forward price curve or on the direct forward price curve according to what is considered suitable. Moreover, it is possible to make the spline more flexible by specifying knot points. Therefore, more difficult curve shapes can be dealt with efficiently. One particularly interesting possibility is to include sine and cosine functions in order to capture cyclicality (or seasonality) that is often encountered in the commodity markets. Furthermore, since the problem is linear, the standard linear regression methods can be used in estimation. By the very nature of the minimization of the squared errors, it is not possible to price market quoted swaps exactly.

• Optimization of Fit Method

The Optimization of Fit method is similar in many ways to the calibration methods used in the option markets. By expressing the swap prices as functions of direct forward prices and adding the condition that the following forward prices are not allowed to deviate too much from the previous forward prices, it is possible to define a well-formed problem that yields basic analytical formulas. The Optimization of Fit method handles many different curve shapes with little difference in goodnessof-fit results, and is therefore a good candidate for applied work and even for trading desks of investment banks. It is also possible to include futures in the optimization problem fairly easily or build the curve in such a way that the shortend of the curve is built using quoted futures prices and the long-end of the curve is build using the Optimization of Fit method. Hence, this method provides much respected flexibility.

Of the four methods (and their variants), two are constructed so that the market prices of swaps are recovered exactly. Therefore, it would be pointless to test the pricing efficiency of these methods; Moreover, the scarcity of data would render out-of-sample tests meaningless. Hence, the pricing efficiency tests are conducted only on the Cubic Spline method and the Optimization of Fit method.

6 Empirical Results

The empirical section of this study concentrates on comparing the pricing errors that result from fitting the forward curve to the data, using the Spline and the Optimization of Fit methods. To keep the scope of the study manageable, only the simple cubic spline is used. It is left for further studies to investigate the effect of using other functional forms, knot points and seasonality. The data used in this study consists of weekly NBSK Risi pulp data covering the period between June 1998 to October 2001 and of monthly Brent oil data covering the period between October 1997 to February 2002. The data is obtained from Nordea Bank. The interest rate data used in estimation consists of Libor rates and zero rates is retrieved from DataStream. The commodity market data is in the form of mid market prices only, so arbitrage violations are not considered in the present study. Figures 9, 10 and 11 show the evolution of the forward curves over the study period for the NBSK Risi pulp data.

For testing estimated forward curves, the fitted price errors are defined as

$$\mathbf{V}_{fx} - \hat{\mathbf{V}}_{fl} \tag{16}$$

That is, the difference between the value of the fixed price leg and the value of the floating price leg. The absolute values of the fitted price errors are then used as a criterion for the goodness-of-fit. Friedman Rank Test is used as a formal test for comparing the relative ranking of the proposed estimation methods.

Summary statistics for NBSK Risi pulp and Brent oil are shown in Tables 1 and 2. Inspection shows very clearly that there are great differences among the estimation methods in their ability to price market swaps. The sum of absolute fitted price errors is very large for the Cubic Spline method without knots. However, for the two other methods, the errors are much less significant, with the Optimization of Fit method





Figure 10: NBSKRISI PULP FORWARD CURVES ESTIMATED USING THE CUBIC SPLINE METHOD (with two equally spaced internal knot points)





Figure 11: NBSKRISI PULP FORWARD CURVES ESTIMATED USING THE OPTIMIZATION OF FIT METHOD

Statistic	Cubic Spline	Cubic Spline with Knots	Optimization of Fit
Mean	235.78	90.38	47.39
Standard Error	12.14	4.54	2.12
Median	199.16	77.67	39.89
Standard Deviation	161.91	60.63	28.30
Kurtosis	2.59	1.04	0.98
Skewness	1.43	1.05	1.04
Range	868.92	326.47	152.36
Minimum	14.22	6.71	5.73
Maximum	883.14	333.18	158.09
Count	178	178	178

Table 1: Summary Statistics NBSK Risi Pulp

- Reported statistics are for the Sum of Absolute Fitted Price Errors.
- Data consists of weekly observations for NBSK Risi Pulp swap quotes from June 1998 to October 2001.
- Models used are: standard Cubic Spline for direct forward prices, standard Cubic Spline for average forward prices with two internal knot points and Optimization of Fit method for direct forward prices.

giving the smallest total errors and standard deviations. Casual inspection immediately suggests that the standard Cubic Spline method without added flexibility is not suitable for estimating the forward curve for commodities.

In order to gain more insight into how the relative portions of the forward curve are fitted, the maturity spectrum is divided into five annual subperiods. The results are reported in Tables 3 and 4. Within these intervals, the absolute fitted price errors are weighted by the respective annuity factors in order to examine the effect on the actual price quotations derived from the analysis. The results indicate, once again, that the Optimization of Fit method yields implied price quotations that are reasonably close to the actual market prices that were used as inputs in the analysis. The standard Cubic Spline method, on the other hand, implies price quotes that are intolerably far from the actual market quotes: The pricing errors range from 50 cents to almost 5 dollars, with

Statistic Brent oil	Cubic Spline	Cubic Spline with Knots	Optimization of Fit
Mean	13.57	4.69	4.58
Standard Error	1.14	0.44	0.43
Median	11.90	3.66	3.91
Standard Deviation	9.06	3.46	3.40
Kurtosis	0.66	0.75	0.44
Skewness	0.99	1.24	1.01
Range	37.17	13.59	14.13
Minimum	1.24	0.94	0.55
Maximum	38.41	14.52	14.68
Count	63	63	63

Table 2: Summary Statistics Brent Oil

- Reported statistics are for the Sum of Absolute Fitted Price Errors.
- Data consists of monthly observations for Brent oil swap quotes from October 1997 to October 2002.
- Models used are: standard Cubic Spline for direct forward prices, standard Cubic Spline for average forward prices with two internal knot points and Optimization of Fit method for direct forward prices.

more severe errors observed at the short-end of the curve.

In general, it seems that the methods yield larger pricing errors at the short-end of the curve than at the long-end of the curve. This phenomenon was also observed by Bliss (1996), who studied the performance of term structure estimation methods to fit the prices of Treasury securities. The most likely reason for this is that the initial swap curves tend to be relatively flat for longer maturities but often very steeply upward or downward sloping for maturities below two years. As the long-end of the curve contains more observations than the short-end, the minimization of the squared errors will put more weight on fitting the long end correctly. A partial remedy for this would be to introduce more knots to the Spline method to make it more flexible in dealing with shorted maturities. The Optimization of Fit method could be made more flexible by making the λ -parameter time dependent, so that the smoothness penalty would have a

Method	0 - 1 year	1 - 2 years	2 - 3 years	3 - 4 years	4 - 5 years
Cubic Spline	4.85	2.19	1.45	0.60	0.68
Cubic Spline w Knots	4.42	0.58	0.47	0.27	0.15
Optimization of Fit	1.88	0.30	0.28	0.08	0.09

Table 3: Non-parametric Test of Mean Absolute Pricing Errors (NBSK Risi)

• Errors are defined as the mean error weighted with the annuity factor within the shown intervals

Table 4: Non-parametric Test of Mean Absolute Pricing Errors (Brent oil)

Method	0 - 1 year	1 - 2 years	2 - 3 years	3 - 4 years	4 - 5 years
Cubic Spline	0.438	0.123	0.069	0.031	0.032
Cubic Spline w Knots	0.246	0.047	0.47	0.008	0.004
Optimization of Fit	0.135	0.042	0.28	0.014	0.010

• Errors are defined as the mean error weighted with the annuity factor within the shown intervals

smaller effect on the short maturities. Of course, each presented method may also utilize simple weighting schemes.

The formal comparison of the relative performance of the methods is carried out using the non-parametric Friedman Rank test applied to the sum of absolute fitted price errors over the whole sample. Results are given in Table 5. For NBSK Risi pulp, the evidence strongly rejects the null of no difference between the methods. The test statistic 336.58 is significant at 1 % level. In fact, the standard Cubic Spline method without knots ranks the worst in all 178 cases, whereas the Optimization of Fit method ranks the best in all but eight cases. For Brent oil, the results are not quite so strong, in particular, there do not seem to be very big differences between the Optimization of Fit method and Spline method with knots. The test statistic for Brent oil data is 95.14, which is still significant at 1 % level. Now, the Optimization of Fit method provides the best fit in 36 cases, whereas the Spline method with knots is the best method in 27

Table 5:	Non-parametric	Test of Mean	Absolute	Pricing	Errors
				· · · · · · · · · · · · · · · · · · ·	

Statistic	Friedman Stat	Cubic Spline	Cubic Spline w Knots	Optimization of Fit
NBSK Risi				
Rank Sum	336.58 **	532	350	186
Brent Oil				
Rank Sum	95.14 **	189	99	90

* (**) indicates that the Friedman Statistic is significant at 5 % (1 %) level

- Reported statistics are for the Absolute Fitted Price Errors.
- Data consists of weekly observations for NBSK Risi Pulp swap quotes from June 1998 to October 2001 and of monthly monthly observations for Brent oil swap quotes from October 1997 to October 2002.
- Models used are: standard Cubic Spline for direct forward prices, standard Cubic Spline for average forward prices with two internal knot points and Optimization of Fit method for direct forward prices.

cases. The standard Cubic method without knots ranks worst in all cases.

In general, the performance of the fitting methods for estimation of the forward curve for NBSK Risi pulp indicates a strong preference for the non-parametric Optimization of Fit method over the Cubic Spline methods. Investigation shows that the short end of the curve is more demanding for the fitting methodologies to handle than the long end. This is true for all methods. From the data perspective, the NBSK Risi pulp data yields more demanding curve shapes overall than Brent oil data.

7 Extensions

This chapter discusses some possible extensions to the presented methods for handling forward curve estimation more flexibly. The most important extension is related to the handling of the futures contracts as part of the curve generation process. There are, in general, two possibilities for doing this: The prices of futures contracts can be taken as an internal part of the optimization problem; alternatively the curve can, at least in some cases, be bootstrapped.

In real trading situations, the market participants have straightforward requirements for suitable forward curve derivation methodologies. First and foremost, the resulting forward curve has to be able to recover the prices of quoted swaps as accurately as possible. Usually, the errors have to be within the bid-ask spread, or else the proposed method is unsuitable and must be improved and extended. The empirical evidence shows that, in general, the Optimization of Fit method does a good job in recovering the market prices of longer term instruments. However, at times, the short end causes problems due to the often observed steep and concave form of the sequential quotes. The method can be modified if a liquid futures market exists by creating the short end of the curve from the futures price quotes and fitting only the long end of the curve using the Optimization of Fit method.

One possible extension would be to introduce Nelson and Siegel (1987) parameterization to the forward curve and investigate how well it would price the market swap quotes in different market scenarios 5 .

8 Conclusions

Several methods for deriving the forward curve from the market prices of commodity swaps are developed. The methods can be classified into 1) bootstrapping methods and 2) fitting methods. By definition, the bootstrapping methods recover the market swap quotes exactly. However, the downside of these methods is the saw-toothed shape of the resulting forward curve. In particular, if the averaging frequency of the underlying commodity swaps is daily, then the bootstrap method will result in a forward curve that is excessively badly shaped. By contrast, the fitting methods are able to generate smooth curves. However, the methods vary in terms of pricing performance, with all fitting methods performing worse than the Bootstrap method for recovering market swap quotes.

In addition to developing methods for forward curve estimation, two fitting methods were tested with data from pulp and oil swap markets. The evidence shows that the Optimization of Fit method is preferred to the Cubic Spline method (even with internal knot points). The Cubic Spline method performs so poorly in recovering the market prices of quoted swaps that its application to real trading situations cannot be considered. The Optimization of Fit method, by contrast, provides a reasonable alternative for traders, though its properties in different curve shape environments and extensions to the method's flexibility need to be explored further.

Extensions to the methodologies presented in this study are suggested. The most important issue is the inclusion of the futures prices into the curve building procedure, which calls for some modifications to the basic structure of the methods. In general, the most versatile way of including the futures contracts is to divide the curve into two parts where the short-end of the curve is built directly from the futures price quotes and the long-end of the curve is derived using the Optimization of Fit method. This approach provides a very appealing alternative for traders, since the liquid part is built from the most liquid instruments and the longer end is fitted to market prices of swaps. Empirical evidence suggests that the Optimization of Fit method performs very well especially in fitting the long maturity swaps. Finally, the paper presents also a simple application of the sine and cosine function in conjunction with the Cubic Spline method to introduce seasonality to the forward curve. The issue of bringing in more economic realism to the estimated forward curve is left for further studies to explore. This realism is particularly important in the energy and agricultural markets where cyclical and seasonal components are generally very strong and identifiable.

Notes

 $^1{\rm The}$ description of the bootstrap method can be found from any standard text book on fixed income markets.

²Some of the most well known traditional models include Brennan and Schwartz (1985), Gibson and Schwartz (1990) and Schwartz (1997), among many others.

³There exist many settlement procedures depending on market conventions and common practices. Mostly, the contracts are settled against an arithmetic average price and sometimes the nearby futures contract is used in place of the spot price in order to calculate the settlement price.

⁴This is to some extent analogous to the interest rate market, where the yields of benchmark bonds cannot readily be used to accurately price other bonds.

⁵This idea was taken up by Jari Käppi during a seminar session at Helsinki School of Economics

References

- Bliss, R. R., 1996, "Testing Term Structure Estimation Methods," Working paper 96-12a, Federal Reserve Bank of Atlanta, November.
- Brennan, M. J., 1958, "The Supply of Storage," American Economic Review, 48, 50–72.
- Brennan, M. J., and E. S. Schwartz, 1985, "Evaluating Natural Resource Investments," Journal of Business, 58, 133–155.
- Chambers, D. R., W. T. Carleton, and D. W. Waldman, 1984, "A New Approach to Estimation of the Term Structure of Interest Rates," *Journal of Financial and Quantitative Analysis*, 19, 233–252.
- Cortazar, G., and E. S. Schwartz, 1994, "The Valuation of Commodity Contingent Claims," *Journal of Derivatives*, 1, 27–39.
- Cox, J. C., J. E. Ingersoll, and S. A. Ross, 1981, "The Relation between Forward Prices and Futures Prices," *Journal of Financial Economics*, 9, 321–346.
- Delbaen, F., and S. Lorimier, 1992, "Estimation of the Yield Curve and the Forward Rate Curve Starting from a Finite Number of Observations," *Insurance: Mathematics* and Economics, 11, 259–269.
- Fama, E. F., and R. R. Bliss, 1987, "The Information in Long-Maturity Forward Rates," American Economic Review, 77, 680–692.
- Fisher, M., D. Nychka, and D. Zervos, 1995, "Fitting the Term Structure of Interest Rates with Smoothing Splines," Working paper 95-1, Federal Reserve Board, Finance and Economics Discussion Series, January.
- Gibson, R., and E. S. Schwartz, 1990, "Stochastic Convenience Yield and Pricing of Oil Contingent Claims," *Journal of Finance*, 45, 959–976.

- Heath, D., R. A. Jarrow, and A. J. Morton, 1992, "Bond Pricing and the Term Structure of Interest Rates: A New Methodology for Contingent Claims Valuation," *Econometrica*, 60, 77–105.
- Hilliard, J. E., and J. Reis, 1998, "Valuation of Commodity Futures and Options under Stochastic Convenience Yields, Interest Rates and Jump Diffusions in the Spot," *Journal of Financial and Quantitative Analysis*, 33, 33–59.
- Käppi, J., 1997, "Essays on Fixed Income Derivatives," Doctoral dissertation, Helsinki School of Economics.
- McCulloch, J. H., 1971, "Measuring the Term Structure of Interest Rates," Journal of Business, 44, 19–31.
- Miltersen, K. R., and E. S. Schwartz, 1998, "Pricing of Options on Commodity Futures with Stochastic Term Structures of Convenience Yields and Interest Rates," *Journal* of Financial and Quantitative Analysis, 33, 33–59.
- Nelson, C. R., and A. F. Siegel, 1987, "Parsimonious Modeling of Yield Curves," Journal of Business, 60, 473–489.
- Press, W. H., S. A. Teukolsky, W. T. Vetterling, and B. P. Flannery, 1992, Numerical Recipes in C: The Art of Scientific Programming. Cambridge University Press, Cambridge, 2 edn.
- Reisman, H., 1991, "Movements of the Term Structure of Commodity Futures and the Pricing of Commodity Claims," Unpublished manuscript, Haifa University, Israel.
- Schwartz, E. S., 1997, "The Stochastic Behavior of Commodity Prices: Implications for Valuation and Hedging," *Journal of Finance*, 52, 923–973.

- Shea, G. S., 1984, "Pitfalls in Smoothing Interest Rate Term Structure Data: Equilibrium Models and Term Structure Data," *Journal of Financial and Quantitative Analysis*, 19, 253–269.
- Suits, D. B., A. Mason, and L. Chan, 1978, "Spline Functions Fitted by Standard Regression Methods," *Review of Economic Studies*, 60, 132–139.
- Vasicek, A. O., and H. G. Fong, 1982, "Term Structure Modeling Using Exponential Splines," *Journal of Finance*, 37, 339–348.

Appendix

Implementing the optimization method requires the calculation of the gradient of the target function, i.e. equation (15). This can be done either numerically or analytically. It transpires that the derivatives of the target function are straightforward to obtain (in the calculations below; $\lambda = 1$).

$$\begin{aligned} &\frac{\partial Fct}{\partial F(t,T_i)} = 2 \times \sum_{j=i+1}^{N} \left(V_{fl}(t,T_j) - V_{fx}(t,T_j) \right) \times \left(\gamma P(t,T_i) + (1-\gamma)P(t,T_{i+1}) \right) \\ &+ 2 \times \left(V_{fl}(t,T_i) - V_{fx}(t,T_i) \right) \times \gamma P(t,T_i) \\ &+ 2 \times \left(F(t,T_i) - F(t,T_{i-1}) \right) - 2 \times \left(F(t,T_{i+1}) - F(t,T_i) \right) \\ &\frac{\partial Fct}{\partial F(t,T_N)} = 2 \times \left(V_{fl}(t,T_N) - V_{fx}(t,T_N) \right) \times \gamma P(t,T_N) \\ &+ 2 \times \left(F(t,T_N) - F(t,T_{N-1}) \right) \end{aligned}$$

where

$$V_{fl}(t, T_i) = \sum_{j=1}^{i} \left(\gamma F(t, T_j) + (1 - \gamma) F(t, T_{j-1}) \right) \times P(t, T_j)$$
$$V_{fx}(t, T_i) = X \sum_{j=1}^{i} P(t, T_j)$$
$$\gamma = (m+1)/(2m)$$

and m is the number of price observations on a settlement period, c.f. linear interpolation constraint. The futures contracts, f_k , can be included in the optimization by defining

$$\frac{\partial f_k}{\partial F(t,T_{i>k})} = 2 \times \tau \times (\tau F(t,T_i) + (1-\tau)F(t,T_{i-1}) - f_k)$$
$$\frac{\partial f_k}{\partial F(t,T_{i$$

where τ is time (in years) from the previous forward price to the settlement date of the futures contract. Subscript k denotes the position of the futures price relative to the forward curve points. Additionally, the futures prices are assumed equal to forward prices. Therefore, the objective is to match the observed futures prices with the optimized forward price curve. As the linear interpolation again is assumed, only the forward prices

right before and after any f_k contribute to the result of the optimization.

Minimization Algorithm:

The derivatives of the target function with respect to the given set of forward prices can be utilized, for example, in the implementation of the method of steepest descent (as is done in this study). Below is a schematic description of the methodology used in this paper. This method is called a *simple gradient search*.

Choose the direction where the value of the target function, f decreases most quickly, which is in the direction opposite to the gradient, $\nabla f(\mathbf{x}_i)$. Start the search at an arbitrary point, or preferably with some reasonable initial value, and then move along the opposite direction of the gradient until the solution is reached (or is close enough). In other words, the iterative procedure is

$$\mathbf{x}_{k+1} = \mathbf{x}_k - \lambda \nabla f(\mathbf{x}_k)$$

where λ can be also be chosen to be of optimal size; for details see Press, Teukolsky, Vetterling, and Flannery (1992). Alternatively, a constant value for λ can be used, then the method is much simpler and requires many more iterations. However, each iteration takes less time to compute and experimentation with the algorithm helps in choosing a suitable step size. Continue these iterations until $|\nabla f(\mathbf{x}_k)| < \epsilon$.

ESSAY 2:

Principal Components Analysis of Commodity Forward Curves: Evidence from Oil and Pulp Swap Markets

Abstract

This paper investigates the factor structure of commodity forward curve dynamics using data from pulp and oil markets. The data used consists of swap contract quotes, from which forward curves are derived using an optimization algorithm. A three factor model explains 89% of the price variation of the oil forward curves and 84% of the price variation of the pulp forward curves. The factor structure, especially in pulp derivatives market, is more complex than found in many other studies conducted mainly using interest rate data. Possible reasons for this phenomenon are discussed. The dynamics of the forward curve is important for practitioners pricing and hedging derivatives contracts and for economists studying stochastic movements of economic variables. Traditionally, modelling of the yield curve movements has been an active area of research, both theoretically and empirically. By contrast, research on commodity forward curve dynamics has been relatively scarce. This paper contributes to the research on commodity forward curve dynamics by presenting empirical evidence on the factor structure of long term forward prices estimated from the broker quotes of the swap contracts.

Analogously to the study of the interest rates, there has been two approaches to the study of commodity prices. The traditional method has focused on modelling the stochastic process for the spot price of the commodity and possibly other state variables, such as the convenience yield. Seminal research along these lines include Brennan and Schwartz (1985), Gibson and Schwartz (1990) and Schwartz (1997). More recently, the study of the stochastic movements in commodity prices has concentrated on modelling the whole term structure of either forward prices directly, Cortazar and Schwartz (1994), or convenience yields, see for example Miltersen and Schwartz (1998). This current paper builds on the whole term structure modelling principle.

Earlier research on the commodity forward curve dynamics has concentrated on modelling the factor structure of the short term contracts, i.e. exchange traded futures. Among the earliest of this research was Cortazar and Schwartz (1994), who studied the dynamics of the futures price curve constructed from copper futures. They found that the factor structure of the copper futures curve was surprisingly similar to the factor structure of yield curve movements. Moreover, the explanatory power of the two most important principal components was 97 percent. By contrast, Litterman and Scheinkman (1988) found that the explanatory power of the two most important principal components of the yield curve movements was remarkably lower, at around 90 percent. Clewlow and Strickland (2000) studied the factor structure of NYMEX oil futures and they found that three factors explained over 98 percent of the variation of the futures price movements during the period between 1998 to 2000. A recent paper by Tolmasky and Hindanov (2002) investigated the dynamics of the petroleum futures contracts. In particular, they focused on isolating the effect of seasonality in the factor structure of returns. They found that, especially for heating oil, seasonality is an important variable driving the factor structure, however it's statistical significance is somewhat unclear. Crude oil and petroleum markets were not found to be affected by seasonality. In a closely related line of research, Koekebakker and Ollmar (2001) studied the forward curve dynamics using data from the Nordpool electricity derivatives exchange. The explanatory powers they report are fairly low in comparison, the most likely reason being the extremely complex dynamics of the electricity spot and forward prices. Koekebakker and Ollmar (2001) use fitted curves as is also done in this current paper.

The results obtained in this study extend the results of the empirical investigations of the forward curve movements by presenting evidence using data from two distinct commodity markets. In particular, the focus is on modelling the dynamics of the long term forward prices that are implied, not actually observed, from the market quotes of the swap contracts. Using the swap market data significantly complicates the analysis and the results obtained are also less clear cut than those obtained using the data on the short term futures contracts. The data used consists of Brent crude oil swap quotes up to five years, covering the period between 1997 to 2002, and NBSK Risi pulp swap quotes, also up to five years, covering the period between 1998 to 2001. The former data is in the form of monthly observations and the latter is weekly sampled. Principal Components Analysis (PCA) applied to the implied forward curve movements reveals complex factor structures and the explanatory power of the first three principal components is around 89 percent for oil data and 84 percent for pulp data. The paper is organized as follows. Section 1 introduces the underlying theoretical model. Section 2 presents and discusses the data used in the study. Section 3 introduces the method used for estimating the forward curves and Section 4 gives a description of the Principal Components Analysis method. In Section 5, the empirical results are discussed and comparisons are made with results obtained in earlier studies. Finally, Section 6 concludes the paper and gives suggestions for further research.

1 Model

The model studied here is similar to the model proposed by Reisman (1991) and Cortazar and Schwartz (1994). The main idea is to model the movements of the forward curve directly, instead of modelling the forward prices as a function of the spot price and convenience yield processes. Utilizing this approach, the principal components analysis is applied to the movements of the whole extracted term structure of the forward prices. The procedure for obtaining the term structures of forward prices is explained in the next section.

In the analysis that follows, the spot price of a commodity, at time t, is denoted by S(t). A futures contract at time t, for delivery at time T, is denoted F(t,T). The underlying market is assumed to be complete and frictionless. In the absence of arbitrage opportunities there exists an equivalent martingale measure \mathbb{Q} , under which all discounted asset prices are martingales (Harrison and Kreps (1979)).¹ This implies that, under the risk-neutral pricing measure, the instantaneous expected return on all financial assets is the instantaneous risk free rate and hence, the expected return on futures contracts is equal to zero. Therefore, the stochastic process for the futures (and forward) prices is given by

$$dF(t,T) = \sum_{j=1}^{K} \sigma_j(t,T)F(t,T)dW_j(t)$$
(1)

and in integrated form

$$F(t,T) = F(0,T) \exp\left(-\frac{1}{2}\sum_{j=1}^{K}\int_{0}^{t}\sigma_{j}^{2}(u,T)du + \sum_{j=1}^{K}\int_{0}^{t}\sigma_{j}(u,T)dW_{j}(u)\right)$$
(2)

where $dW_1, dW_2, ..., dW_K$ are independent increments of Brownian motions under the risk-neutral measure, and $\sigma_j(t, T)$'s are the volatility functions of the futures (or forward) prices. The volatilities are only functions of time to maturity (T - t). The analysis presented here is based on the assumption of constant interest rates, and therefore, the futures prices and forward prices are equivalent as shown by Cox, Ingersoll, and Ross (1981).

2 Data

The forward price model (1) describes the stochastic evolution of each of the forward prices along the forward curve under an equivalent martingale measure. However, observations are taken under the real world measure. This is not a problem, since only volatility functions are of interest here, and they are invariant with respect to the measure change². Let $F(t_i, T_j)$ denote the forward price at time t_i , with maturity T_j (t < T), for all i = 1, 2, ...N and j = 1, 2, ...M. The instantaneous proportional change in the forward price is approximated by

$$\frac{dF(t_i, T_j)}{F(t_i, T_j)} = \frac{F(t_i, T_j) - F(t_{i-1}, T_j)}{F(t_{i-1}, T_j)} = x_{i,j}$$
(3)
The data set $\mathbf{X}_{(N,M)}$ is a matrix of returns (3) constructed from the fitted forward price curve and swap price curve

$$\mathbf{X}_{(N,M)} = \begin{pmatrix} x_{1,1} & x_{1,2} & \dots & x_{1,M} \\ x_{2,1} & x_{2,2} & \dots & x_{2,M} \\ \vdots & \vdots & \ddots & \vdots \\ x_{N,1} & x_{N,2} & \dots & x_{N,M} \end{pmatrix}$$
(4)

The matrix entries are obtained in the following way. First, the forward curves are derived from the swap price data using Järvinen (2002). The forward curves are then used to find the individual forward price observations. From these observations, the percentage differences are calculated. This procedure is replicated until the whole matrix is filled.

The data used in this study consists of weekly NBSK Risi pulp data covering the period between June 1998 to October 2001 and of monthly oil data (European Brent) covering the period between February 1997 to February 2002. The data was obtained from Nordea Bank Finland. The interest rate data used in estimation consists of Libor rates and zero rates. The European interest rate data consists of EuroLibor quotes and zero rates until 1.1.1999; from that date onwards the data consists of Euribor quotes and zero rates. The interest rate data was retrieved from DataStream. The commodity data consists of mid market quotes. Tables 1 and 2 report the summary statistics for NBSK Risi pulp forwards and swaps, and Tables 3 and 4 report the summary statistics for Brent oil forwards and swaps.

Statistic	Index	$6 \mathrm{m}$	1y	2y	3y	5y
Mean	-0.0009	-0.0004	-0.0004	-0.0004	-0.0004	0.0000
Standard Deviation	0.0167	0.0109	0.0116	0.0098	0.0109	0.0184
Kurtosis	9.2818	1.9003	2.2834	1.7928	1.5060	4.7249
Skewness	-0.4417	0.6824	0.4424	0.0452	0.3460	0.2415
Range	0.1495	0.0688	0.0778	0.0646	0.0656	0.1514
Minimum	-0.0755	-0.0306	-0.0327	-0.0324	-0.0305	-0.0555
Maximum	0.0741	0.0382	0.0451	0.0323	0.0351	0.0958
Count	177	177	177	177	177	177

Table 1: NBSK Risi Forwards Summary Statistics

Table 2: NBSK Risi Swaps Summary Statistics

Statistic	Index	$6 \mathrm{m}$	1y	2y	3y	5y
Mean	-0.0009	-0.0006	-0.0005	-0.0005	-0.0005	-0.0004
Standard Deviation	0.0167	0.0099	0.0101	0.0080	0.0073	0.0059
Kurtosis	9.2818	1.8828	2.4779	1.7625	0.9910	2.2131
Skewness	-0.4417	-0.2175	0.4850	0.0794	-0.0438	0.3648
Range	0.1495	0.0626	0.0672	0.0537	0.0382	0.0343
Minimum	-0.0755	-0.0333	-0.0289	-0.0279	-0.0188	-0.0169
Maximum	0.0741	0.0293	0.0383	0.0258	0.0194	0.0175
Count	177	177	177	177	177	177

Table 3: Brent Forwards Summary Statistics

Statistic	Index	6m	1y	2y	Зy	5y
Mean	0.0062	0.0024	0.0021	0.0051	0.0025	0.0008
Standard Deviation	0.1255	0.0650	0.0578	0.0547	0.0491	0.0630
Kurtosis	0.5542	0.4751	1.7357	1.3674	0.4552	0.5037
Skewness	0.3486	-0.1424	0.6150	0.5294	0.1060	0.5465
Range	0.6924	0.3386	0.3400	0.2950	0.2471	0.3165
Minimum	-0.3059	-0.1754	-0.1299	-0.1239	-0.1235	-0.1340
Maximum	0.3866	0.1631	0.2101	0.1712	0.1236	0.1825
Count	62	62	62	62	62	62

Statistic	Index	6m	1y	2y	Зy	5y
Mean	0.0062	0.0031	0.0018	0.0023	0.0024	0.0016
Standard Deviation	0.1255	0.0709	0.0645	0.0434	0.0383	0.0390
Kurtosis	0.5542	1.3295	1.0097	0.3944	0.3283	0.4677
Skewness	0.3486	0.2551	-0.4186	0.0352	0.0610	0.1655
Range	0.6924	0.4248	0.3501	0.2161	0.1887	0.1889
Minimum	-0.3059	-0.1959	-0.2063	-0.1099	-0.0846	-0.0859
Maximum	0.3866	0.2290	0.1439	0.1062	0.1041	0.1030
Count	62	62	62	62	62	62

Table 4: Brent Swaps Summary Statistics

3 Curve Estimation

The first step in applying the principal analysis to the forward curve movements, is to estimate the forward curves from the par swap quotes. To introduce the idea, we consider a standard commodity swap contract, with maturity T_N and a fixed contract price $G(t, T_N)$. The present value of this fixed leg of the swap, given the discount bond prices P(t, T) is

$$V_{fx}^{N} = G(t, T_{N}) \sum_{i=1}^{N} P(t, T_{i})$$
(5)

i.e. a simple sum of the present values of the payments. The other side of the swap contract, the floating leg, is composed of the present values of the unknown forward prices of the commodity at settlement dates, i = 1, 2, ...N. That is,

$$V_{fl}^{N} = \sum_{i=1}^{N} F(t, T_i) P(t, T_i)$$
(6)

where forward prices may be based on average or single observation settlement values. If we know the forward price curve, the par swap prices for all maturities, $t \ge T$, can be obtained by setting the values of the floating and fixed legs equal. Solving for the par swap price $G(t, T_N)$ gives

$$G(t, T_N) = \frac{\sum_{i=1}^{N} F(t, T_i) P(t, T_i)}{\sum_{i=1}^{N} P(t, T_i)}$$
(7)

which shows that the par swap price is defined as the ratio of the discounted value of the floating price payments and an annuity. Therefore, analogously with interest rate swaps, the par swap price can be interpreted as a present value weighted sum of forward prices.

The method for extracting F(t, T)'s is based on the Optimization of Fit method, introduced by Järvinen (2002). The method is applied by minimizing the sum of squared pricing errors, which is an analogous idea to the option model calibration to observed implied volatilities. The aim is to fit the observed swap prices as precisely as possibly while maintaining a reasonable degree of smoothness in the output forward curve. The target function of the minimization problem is

$$\min_{\{F_i\}_{i=1}^N} \left[\sum_{i=1}^N \left(V_{fl}(t, T_i) - V_{fx}(t, T_i) \right)^2 + \lambda \sum_{i=1}^N \left(F(t, T_i) - F(t, T_{i-1}) \right)^2 \right]$$
(8)

where parameter λ is introduced to adjust the weight put on the forward curve smoothness. The forward prices that are the solution of the problem are direct forward prices, and not average-based. ³ For further of this procedure, see Järvinen (2002). The optimization problem can be solved using any suitable algorithm. The method utilized in this paper is based on a simple gradient search using analytic derivatives. For explanation on the various optimization algorithms, see Press, Teukolsky, Vetterling, and Flannery (1992).

In choosing λ , two points need to be mentioned. First, if one picks $\lambda = 0$, then the result of the optimization will converge to the bootstrapped curve and the target function value will be equal to zero. In other words, an exact match can be obtained using this method. On the other hand, if a high enough λ is chosen, then the final forward curve will be flat, starting from the spot value S(t). Obviously, a reasonable value from λ is such that a smooth, yet reasonably shaped and correctly pricing forward curve will result. For the purposes of this study, λ is set equal to one. The decision is based on experimenting with the algorithm, not explicitly on any measurable criteria.

4 Principal Components Analysis

Principal Components Analysis (PCA) is a statistical tool used to identify a structure within a set of interrelated variables. Applying (PCA) to the data, the number of orthogonal factors and the corresponding volatility coefficients (assuming that volatility functions depend only on the time to maturity (T-t)) can be estimated. The data consists of N observations of M different variables, i.e. percentage differences calculated from curve points. Hence, the data matrix is given by (4).

The sample covariance matrix is of order M, and is denoted by ω . The orthogonal decomposition of the covariance matrix is given by

$$\mathbf{\Omega} = \mathbf{C}\mathbf{\Lambda}\mathbf{C}'\tag{9}$$

where

$$\mathbf{C} = [\mathbf{c}_{1}, \mathbf{c}_{2} \dots \mathbf{c}_{M}] = \begin{pmatrix} c_{1,1} & c_{1,2} & \dots & c_{1,M} \\ c_{2,1} & c_{2,2} & \dots & c_{2,M} \\ \vdots & \vdots & \ddots & \vdots \\ c_{N,1} & c_{N,2} & \dots & c_{N,M} \end{pmatrix}$$
(10)

and

$$\mathbf{\Lambda} = \begin{pmatrix} \lambda_{1,1} & 0 & \dots & 0 \\ 0 & \lambda_{2,2} & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & \lambda_{M,M} \end{pmatrix}$$
(11)

The Λ -matrix is a diagonal matrix and the diagonal elements are the eigenvalues $\lambda_{1,1}, \lambda_{2,2}, ..., \lambda_{M,M}$. **C** is an orthogonal matrix of order M. The columns of **C** are the eigenvectors corresponding to $\lambda_{j,j}$. **C'** denotes the transpose of **C**. The matrix $\mathbf{P} = \mathbf{X}\mathbf{C}$ is the matrix of principal components. The columns \mathbf{p}_j of **P** are linear combinations of the columns of **X**, i.e. the j:th principal component is

$$\mathbf{p}_{\mathbf{j}} = \mathbf{X}\mathbf{c}_{\mathbf{j}} = \mathbf{x}_{\mathbf{1}}c_{1j} + \mathbf{x}_{\mathbf{2}}c_{2j} + \ldots + \mathbf{x}_{\mathbf{M}}c_{Mj}$$
(12)

In order to explain all the variance in the sample, one needs to use all M principal components. However, the main idea behind using (PCA) is to reduce the dimensionality of the data. Therefore, the covariance structure is approximated by using K < M largest eigenvalues. The larger the proportion of the explained variance, the better the objective is achieved. The criteria for selecting K is somewhat ambiguous, since there exists no clearcut statistical criterion for selecting significant eigenvalues, and hence, the number of factors ⁴. Therefore the conventional methodology, employed in the majority of the finance literature, is to add factors until the cumulative explained variance reaches a specified limit; this procedure is also used in this study. In addition, financial derivatives multifactor models usually aim at constructing a model that contains from two to four driving factors. The proportion of the total sample variance explained can be found using the following formula

$$CEV_K = \frac{\sum_{j=1}^K \lambda_j}{\sum_{j=1}^M \lambda_M}$$
(13)

Where CEV_K denotes Cumulative explained variance of first K factors.

	Index	6y	1y	2y	3y	5y
Index	1	-0.01	-0.09	-0.06	-0.06	0.02
6m	-0.01	1	0.85	0.06	-0.01	0.01
1y	-0.09	0.85	1	0.17	-0.05	-0.12
2y	-0.06	0.06	0.17	1	0.64	-0.36
3y	-0.06	-0.01	-0.05	0.64	1	-0.03
5y	0.02	0.01	-0.12	-0.36	-0.03	1

Table 5: NBSK Risi Fwds Correlations

Table 6: NBSK Risi Swap Correlations

	Index	6m	1y	2y	3y	5y
Index	1	0.48	0.23	0.15	0.07	0.07
6m	0.48	1	0.62	0.62	0.4	0.47
1y	0.23	0.62	1	0.83	0.6	0.51
2y	0.15	0.62	0.83	1	0.73	0.6
3y	0.07	0.4	0.6	0.73	1	0.67
5y	0.07	0.47	0.51	0.6	0.67	1

5 Empirical Results

Tables 9 and 10 show the results from the PCA analysis applied to both data sets. Empirical results are markedly different, a one factor model is able to explain 62 percent of the variation of returns of the Brent forward prices, whereas it can explain only 38 percent of the returns in the case of NBSK Risi prices. The explanatory power of one factor model, in particular for NBSK Risi pulp, is very low. The most likely reason for this is the stickiness of the reference index price. NBSK Risi index prices are published only monthly, whereas the swap quotes change, in principle, daily, and in most cases at least weekly. Inspection of the empirical correlation matrices (see Tables 5 and 6) reveal that movements of the NBSK Risi pulp prices are almost non correlated (particularly in the short term) whereas correlations of oil price movements are clearly stronger (Tables 7 and 8).

	Index	6m	1y	2y	3y	5y
Index	1	0.36	0.14	-0.25	-0.13	-0.13
6m	0.36	1	0.84	-0.03	0.08	-0.02
1y	0.14	0.84	1	0.26	0.33	0.17
2y	-0.25	-0.03	0.26	1	0.86	0.71
3y	-0.13	0.08	0.33	0.86	1	0.91
5y	-0.13	-0.02	0.17	0.71	0.91	1

Table 7: Brent Fwd Correlations

Table 8: Brent Swap Correlations

	Index	6m	1y	2y	3y	5y
Index	1	0.78	0.63	0.47	0.3	0.16
$6 \mathrm{m}$	0.78	1	0.87	0.74	0.53	0.33
1y	0.63	0.87	1	0.88	0.68	0.46
2y	0.47	0.74	0.88	1	0.93	0.77
3y	0.3	0.53	0.68	0.93	1	0.93
5y	0.16	0.33	0.46	0.77	0.93	1

If more factors are used, then the difference in explanatory power of PCA decreases considerably. A two factor model is able to explain 81 and 63 percent of variation in price returns and a three factor model can explain 89 and 84 percent. Using the eigenvalue criterion that only eigenvalues greater than one are considered significant, the analysis would select a three factor model for Brent and a four factor model for NBSK Risi pulp. For practical modelling purposes, 90 percent has often been considered a minimal threshold. If this criteria is used, then both data sets need a four factor model to explain enough variation of returns. The plotting of eigenvalues for NBSK Risi pulp and Brent oil forward prices are shown in Figures 1 and 2.

Inspection of the factor loadings, see Figures 3 and 4 (Tables 11 and 12), provides more insight on how the dynamics of the forward curves are determined. The most striking finding of the analysis is that contrary to the factor structure found in many other

Eigenval	Variance (% total)	Cumul Eigenval	Cumul $\%$
7.610	38.051	7.610	38.051
5.001	25.006	12.611	63.057
4.278	21.390	16.889	84.447
1.656	8.280	18.545	92.727
0.669	3.346	19.215	96.073
0.108	0.539	19.322	96.612
0.058	0.289	19.380	96.901
0.048	0.240	19.428	97.141
0.048	0.240	19.476	97.380
0.048	0.238	19.524	97.619

Table 9: NBSK Risi Fwd Eigenvalues

Table 10: Brent Fwd Eigenvalues

Eigenval	Variance (% total)	Cumul Eigenval	Cumul $\%$
12.378	61.888	12.378	61.888
3.842	19.210	16.220	81.098
1.627	8.135	17.847	89.233
0.932	4.659	18.778	93.892
0.330	1.651	19.108	95.542
0.217	1.083	19.325	96.625
0.053	0.266	19.378	96.891
0.050	0.248	19.428	97.139
0.048	0.241	19.476	97.380
0.048	0.239	19.524	97.619





Figure 2: Plot of Eigenvalues (Brent Fwd)



studies, see for example Litterman and Scheinkman (1988) and Cortazar and Schwartz (1994), there seems to be no obvious level, slope and curvature factors. By contrast, the factors obtained here exhibit much more complex shapes. There is no general factor that would change the forward prices equally across the whole term structure. For example, in the case of NBSK Risi, a shock to the first factor changes the very short term forward prices upwards, albeit marginally, and the medium term forward prices downwards and finally, the long term forward prices upwards. An interesting note is also that only the first factor is important in explaining the variation in the very long forward prices whereas all four factors have an important impact in explaining the movements in the short end of the forward curve. The conclusion from the analysis of the factor loadings is that when long term contracts are included in the estimation of the dynamics of the forward curve, the complexity of curve changes increases dramatically. Possible explanations for these results could be the strongly mean-reverting nature of the commodity prices and the complex interplay between future demand and supply expectations and also the possibility of storing these commodities for speculating or hedging. Another possibility is that there are great differences in the liquidity of the contracts of different maturities, and that the reference index value changes non-synchronously with the prices of the swap contracts. It is likely that both these factors have an effect on the estimation results.

	Factor 1	Factor 2	Factor 3	Factor 4
3 month	0.1286	0.4087	-0.4804	-0.3098
6 month	0.0461	0.5391	-0.6561	-0.4472
9 month	-0.0444	0.5104	-0.6815	-0.4333
12 month	-0.0992	0.5437	-0.7116	-0.2477
15 month	-0.1437	0.6316	-0.6914	0.1920
18 month	-0.1907	0.6210	-0.4666	0.5726
21 month	-0.3033	0.6493	-0.1704	0.6459
24 month	-0.4914	0.7227	0.2408	0.3678
27 month	-0.5327	0.6310	0.5266	-0.0289
30 month	-0.4893	0.5516	0.6181	-0.1880
33 month	-0.4078	0.5487	0.6616	-0.2374
36 month	-0.1616	0.6409	0.6822	-0.2434
39 month	0.4938	0.6775	0.4853	-0.1397
42 month	0.8639	0.4303	0.1604	0.0042
45 month	0.9337	0.2831	0.0601	0.0524
48 month	0.9520	0.2124	0.0559	0.0628
51 month	0.9592	0.1691	0.0739	0.0635
54 month	0.9624	0.1388	0.0891	0.0630
57 month	0.9639	0.1200	0.0967	0.0630
60 month	0.9643	0.1135	0.0987	0.0632
Expl.Var	7.6102	5.0011	4.2780	1.6561
Prp.Totl	0.3805	0.2501	0.2139	0.0828

Table 11: NBSK Risi Fwd Factor Loadings

	Factor 1	Factor 2	Factor 3	Factor 4
3 month	-0.1198	-0.5095	0.5732	-0.5889
6 month	-0.0133	-0.8371	0.3798	-0.3023
9 month	0.1552	-0.9329	0.0182	0.1437
12 month	0.2625	-0.8840	-0.1496	0.3018
15 month	0.4091	-0.8136	-0.2359	0.2296
18 month	0.6779	-0.5946	-0.3486	-0.0340
21 month	0.8166	-0.1628	-0.4142	-0.2615
24 month	0.8263	0.0900	-0.3969	-0.2973
27 month	0.8764	0.1521	-0.3237	-0.2454
30 month	0.9237	0.1429	-0.1991	-0.1520
33 month	0.9266	0.1518	-0.0978	-0.0732
36 month	0.9329	0.1509	-0.0256	-0.0258
39 month	0.9578	0.0879	0.0582	0.0076
42 month	0.9644	-0.0006	0.1500	0.0383
45 month	0.9511	-0.0266	0.2126	0.0646
48 month	0.9389	0.0060	0.2484	0.0873
51 month	0.9280	0.0636	0.2702	0.1076
54 month	0.9153	0.1209	0.2859	0.1252
57 month	0.9023	0.1649	0.2982	0.1391
60 month	0.8926	0.1905	0.3068	0.1480
Expl.Var	12.3776	3.8420	1.6270	0.9317
Prp.Totl	0.6189	0.1921	0.0814	0.0466

Table 12: Brent Fwd Factor Loadings



Figure 3: Plot of Factor Loadings (NBSK Risi Fwd)

Figure 4: Plot of Factor Loadings (Brent Fwd)



6 Conclusions

This paper has investigated the dynamics of the commodity forward price curves using the principal components analysis on the Brent oil and NBSK Risi pulp data. The forward curves have been estimated using the fitting method, presented in Järvinen (2002). The data consists of broker swap quotes from 1998 to 2002 for Brent and from 1997 to 2002 for NBSK Risi. The longest maturity swaps are five years. The principal component analysis was conducted on the percentage changes of the forward prices from the fitted forward curves.

The main findings of the study are: At least three and even four factors are needed in order to adequately model the dynamics of the forward curves for both data series. Interestingly, the factor structures of the markets analyzed are markedly different, and furthermore do not resemble the structures found in earlier studies: level, slope and curvature. By contrast, the results derived in this paper reveal that the short-end of the curve is very demanding from the modelling perspective. In contrast, the long-end of the curve exhibited simpler dynamics. Possible reasons for these findings include: complex demand, supply and storage dynamics leading to non-synchronously moving forward prices, in particular, in the short-end of the curve; liquidity and reliability of the swap quotes from which the forward curves are derived; and stickiness of the reference index quotes that provide the pseudo spot index value for the fitting algorithm. Finally, the results are naturally dependent on the forward curve estimation algorithm. Since the forward curve points cannot be estimated uniquely from the swap data, a fitting method is necessary in order to obtain the forward curves.

This paper provides the first documented results of applying PCA to model the dynamics of the commodity forward curves using long term swap contract data. Earlier studies have used readily available futures data. As the OTC market for long term commodity derivative contracts is growing rapidly, more research is needed in order to investigate the statistical behavior of the prices of these contracts. As new data arrives at an expanding rate, this provides the academic community fruitful opportunities to investigate the subject further. Moreover, since the futures markets are often very liquid and usually proxy the spot price better than reference index values, future studies might benefit from integrating these two markets in order to study the term structure dynamics.

Notes

¹The equivalent martingale measure is commonly called the risk-neutral measure, in particular in the applied research.

²This is strictly correct only in continuous time

³By direct forward prices we mean forward prices that apply to single settlement date values. On the contrary, average forward prices contain resets from the start of the forward period to the end of the forward period.

⁴Criteria discussed in the literature includes: 1) Scree plot test: the test is carried out by graphical inspection of the eigenvalue plot; eigenvalues are added until the plot levels off. 2) Eigenvalue criterion: eigenvalues that are greater than one are considered significant.

References

- Brennan, M. J., and E. S. Schwartz, 1985, "Evaluating Natural Resource Investments," Journal of Business, 58, 133–155.
- Clewlow, L., and C. Strickland, 2000, Energy Derivatives: Pricing and Risk Management. Lacima Publications, London, England, 1 edn.
- Cortazar, G., and E. S. Schwartz, 1994, "The Valuation of Commodity Contingent Claims," *Journal of Derivatives*, 1, 27–39.
- Cox, J. C., J. E. Ingersoll, and S. A. Ross, 1981, "The Relation Between Forward Prices and Futures Prices," *Journal of Financial Economics*, 9, 321–346.
- Gibson, R., and E. S. Schwartz, 1990, "Stochastic Convenience Yield and Pricing of Oil Contingent Claims," *Journal of Finance*, 45, 959–976.
- Harrison, J. M., and D. M. Kreps, 1979, "Martingales and Arbitrage in Multiperiod Security Markets," *Journal of Economic Theory*, 20, 381–408.
- Järvinen, S., 2002, "Estimating the Forward Curve for Commodities," Unpublished manuscript, Helsinki School of Economics.
- Koekebakker, S., and F. Ollmar, 2001, "Forward Curve Dynamics in the Nordic Electricity Market," Unpublished manuscript, Agder University College.
- Litterman, R., and J. Scheinkman, 1988, "Common Factors Affecting Bond Returns," Journal of Fixed Income, 1, 54–61.
- Miltersen, K. R., and E. S. Schwartz, 1998, "Pricing of Options on Commodity Futures with Stochastic Term Structures of Convenience Yields and Interest Rates," *Journal* of Financial and Quantitative Analysis, 33, 33–59.

- Press, W. H., S. A. Teukolsky, W. T. Vetterling, and B. P. Flannery, 1992, Numerical Recipes in C: The Art of Scientific Programming. Cambridge University Press, Cambridge, 2 edn.
- Reisman, H., 1991, "Movements of the Term Structure of Commodity Futures and the Pricing of Commodity Claims," Unpublished manuscript, Haifa University, Israel.
- Schwartz, E. S., 1997, "The Stochastic Behavior of Commodity Prices: Implications for Valuation and Hedging," *Journal of Finance*, 52, 923–973.
- Tolmasky, C., and D. Hindanov, 2002, "Principal Components Analysis for Correlated Curves and Seasonal Commodities: The Case of the Petroleum Market," *Journal of Futures Markets*, 22, 1019–1035.

ESSAY 3:

Simple Binomial Model for Bermudan Swaptions

Abstract

This paper presents a model for pricing Bermudan swaptions in a standard binomial tree. In particular, we adopt the equal probability specification of Jarrow and Rudd (1983). The model is extremely simple and efficient to implement, but the most obvious gain comes from its intuitive nature and its ability to incorporate market information without calibration. The methodology involves the computation of recombining Markov lattices for every floating leg underlying the swaption. An option tree is constructed and used to value a Bermudan swaption. We give examples of applying the model to price interest rate and commodity swaptions.

A BERMUDAN SWAPTION is an option which, at each predetermined date in the exercise schedule, gives the owner the right to enter into a swap, either to pay fixed (payer swaption) or receive fixed (receiver swaption). The importance of Bermudan type swaptions arises from the widespread use of call features in the issuance of corporate and government bonds. Bermudan swaptions provide hedges for these bonds and the Bermudan swaption pricing models can also be used to estimate the value of the call provision for the benefit of both the issuer and the investor.

The most popular models for pricing Bermudan swaptions are low-dimensional (typically one-factor, sometimes two factors). Familiar models include classics such as BDT Black, Derman, and Toy (1990) and HW Hull and White (1990) among many others. The popularity of these well-known one-factor models comes from their low-cost implementation in the binomial (or trinomial) lattices. Moreover, lattices are generally preferred to Monte Carlo methods in the pricing tasks that involve estimation of the early-exercise premium, since using lattices this can be done in a straightforward fashion ¹. It is often claimed that the convenience of one-factor models comes at a cost, since we are restricted to evolve the whole spectrum of rates using only one driving Brownian motion. This severely limits the possible shapes that the yield curve may obtain in the future and also forces the yields to move in tandem. Whether this is a practically relevant, restrictive feature or not is still to be proved. Empirical evidence is, to date, weak and mixed.

In the implementation of the models above, the BDT model comes closest to the market practice, since under this model the interest rates follow the lognormal law. However, there is a drawback: The nodes in the binomial lattice have to be solved numerically, since analytical formulae do not exist 2 . Furthermore, it is not enough to build the tree until option expiry. The tree has to be evolved up until the maturity of the underlying swap. By contrast, the HW model is analytically more tractable, but problems arise due to underlying Gaussian law for the rates. The normal distribution of the interest rates is not what the market assumes and, moreover, it follows that the rates can be negative at positive probability in the HW model. Finally, the calibration to market data is less transparent than in the BDT model, since the volatilities of the rates in HW are absolute, not proportionate to the level of rates as the market implicitly assumes.

To introduce a more realistic framework for the modeling of yield curves, many authors have proposed the so-called market models Miltersen, Sandmann, and Sondermann (1997) and Brace, Gatarek, and Musiela (1997), LMM's (Libor Market Models) and their implementation specific variations. These models describe the evolution of the market quantities (Libor rates), rather than the instantaneous rates of preceding models. The change of modelling standpoint allows for lognormal forward Libor rates. This assumption coincides with the market practice of pricing caps. Therefore, interest rate caps can be priced using an analytical formula and, hence, calibration to market data is straightforward. The purpose of the LMM's is to model the yield curve in a consistent way, using multiple factors to better capture imperfect correlation properties of forward rates. The inclusion of many factors makes it harder to implement tree methods. Therefore, Monte Carlo techniques must be used instead. This field of research has grown tremendously during the past decade, see for example Broadie and Glasserman (1997), Barraquand and Martineau (1995), Carr and Yang (1997), and Longstaff and Schwartz (2001) for various methods to incorporate the value of early exercise into the Monte Carlo method. One of the major motivations behind the popularity of this area of research has been to enable pricing of Bermudan swaptions in the LMM's, see Andersen (1999) and Pedersen (1999).

While the field of yield curve modelling continues to flourish, there still exists the problem of finding a reliable way to model the curve evolution. The empirical evidence has been discouraging and the intuition of the model is easily lost if the model contains multiple driving factors. While multifactor models are intuitively the hardest models to understand and use in practise, the one-factor models have their own problems too. The first and obvious one is the calibration. The HW model for example is Gaussian and must be fitted to a lognormal market, using the parameters of the stochastic differential describing the motion of the hypothetical short rate. BDT, on the other hand, is less analytical and more market oriented.

The method presented in this paper is based on a simple idea of pricing, in an arbitrage-free manner, the European swaptions underlying the Bermudan swaption. Each of the Europeans will be priced using their individual, broker quoted, volatilities. The forward swap rates are recovered exactly from the swap curve. Hence, there is no need for any kind of calibration whatsoever. Moreover, from an intuitive point of view, using the European swaption volatilities as inputs in the pricing along with the forward swap rates means the trader will have a greater degree of trust in the pricing procedure. The pricing procedure goes as follows: first we build trees of underlying forward floating legs of the swap, using the volatilities of the European swaptions. In fact, we only need to produce the end node values. After that we can compute the Bermudan option tree in a standard way of working from back to the root and checking for early exercise. The pricing method is very efficient, since along with the avoidance of any calibration, we can use a lattice similar to the famous Cox, Ross, and Rubinstein (1979) or Jarrow and Rudd (1983), the latter being more suitable for this type of implementation. In addition to presenting the algorithm for standard interest rate Bermudan swaptions, we extend the analysis to cover the case of the commodity Bermudan swaption. For applications of the swap market models, Jamshidian (1997) has proposed pricing Bermudan swaptions using an ordinary binomial lattice in order to get quick and accurate results. He does not, however, present details of the implementation.

The rest of this paper is organized as follows: Section 2 introduces the notation, the general ideas of the swap derivatives and the standard binomial pricing model. In Section 3, we develop the model and show how the pricing of the Bermudan swaptions can be done in this framework. Section 4 contains numerical examples of pricing and hedging of a Bermudan swaption within the model framework and compares the results with the BDT prices. In Section 5, an application to handle the commodity Bermudan swaptions is presented. Section 6 concludes the paper and suggests further research.

1 Notation, Swap Derivatives and Binomial Pricing

In this section, we review the fundamental background of swap and swap derivatives. We also present an application of the famous Cox, Ross, and Rubinstein (1979) (CRR) type binomial tree for the pricing of European swaptions. This will provide us with a tool for extending the model to price Bermudan type swaptions. We start by introducing the basic notation and concepts of swap derivatives, concentrating first on the interest rate swaption.

The fundamental building blocks of the interest rate derivatives are zero coupon bonds. The value at time t of a zero coupon bond maturing at T_i is denoted by $P(t, T_i)$. It is assumed that the zero coupon bond exists for every maturity T_i . The par swap rate is a fixed interest rate that equalizes the present values of the floating and fixed legs. The value of the floating leg at time t is given by

$$Fl_t = P(t, T_0) - P(t, T_N) \tag{1}$$

where the subscripts 0 and N denote the start date and the end date of the contract. The derivation of this result is straightforward and can be found in Rebonato (1996). The value of the fixed leg at time t is the value of an annuity with tenor structure τ^{-3} . In mathematical notation, this is expressed as

$$Fx_t = X\tau \sum_{i=1}^{N} P(t, T_i)$$
⁽²⁾

where X is the fixed rate of interest. Hence, we write the par (spot or forward) swap rate as

$$S_t = \frac{P(t, T_0) - P(t, T_N)}{\tau \sum_{i=1}^{N} P(t, T_i)}$$
(3)

So, the derivation of the values of the floating and fixed legs of any swap at any time t requires only knowledge of the discount function. Similarly, any forward swap rate is instantaneously recovered from the discount function. Note that in the calculation of the spot swap rate, the numerator becomes $P(t,t) - P(t,T_N) = 1 - P(t,T_N)$. The above definitions show that the fundamental quantities in the swap market are very easy to derive. The market practice for extracting the P's is to bootstrap the par swap curve to get the zero coupon bonds for maturities 1, 2, 3 etc years. The intermediate zero coupon prices can be obtained via some interpolation method.

1.1 Standard European Swaption Formula

The underlying model used in this paper is a version of the famous Black (1976) formula applied to swaptions. In this formula, the floating leg of the swap is treated as a stochastic variable whereas the value of the fixed leg is a constant annuity (the discounting factor). The present value of basis point PVBP $\tau \sum_{i=1}^{N} P(t, T_i)$ is used as the martingale measure. Therefore, the floating leg discounted using this numeraire (i.e. the forward swap rate), will be a martingale. The stochastic differential of the forward swap rate, under the forward swap measure \mathbb{Q} , is

$$dS_t = \sigma S_t d\tilde{W}_t \tag{4}$$

where $d\tilde{W}_t$ is the standard Brownian motion under \mathbb{Q} . Hence, the European swaption (payer) price is given by the following well-known formula

$$Payer_{t} = Fl_{t}\mathbf{N}\left(h_{1}\right) - Fx_{t}\mathbf{N}\left(h_{2}\right)$$

$$\tag{5}$$

where

$$h_{1} = \frac{\ln (Fl_{t}/Fx_{t}) + \frac{1}{2}\sigma^{2}T}{\sigma\sqrt{T}}, h_{2} = h_{1} - \sigma\sqrt{T}$$

and $\mathbf{N}(\cdot)$ represents the cumulative standard normal distribution function. In this form, the European payer swaption formula has the most convenient representation. The formula for the receiver swaption can be obtained directly from the put-call parity. The Black (1976) model for European commodity swaptions is derived in Järvinen and Toivonen (2002).

1.2 Binomial Approximation

We now illustrate how the standard binomial model of Cox, Ross, and Rubinstein (1979) is used for calculating the price of a European swaption. We start by defining the riskneutral probability for a futures market.

$$q = \frac{1-d}{u-d} = \frac{1-e^{-\sigma\sqrt{d}t}}{e^{\sigma\sqrt{d}t} - e^{-\sigma\sqrt{d}t}}$$
(6)

where σ is the volatility of the forward swap rate and dt equals time to maturity of the swaption divided by the number of periods in the binomial tree. Therefore, the floating leg of the swap evolves as follows



The requirement u > 1 > d is enough to guarantee that the model is arbitrage-free. To enable calculations of option prices using the model, we construct a binomial tree for the value of the floating leg as follows





In the binomial model above, index i runs from 0 to N (number of periods in the tree). It is a well-known fact that in order to price American options, we must calculate option values in every node to find out whether it pays off to exercise early or not (the optimal stopping problem). However, in the case of European options, the above tree reduces to the calculation of the probability-weighted average of the option values at the end nodes. The formula for European payer swaption is

$$Payer_{t} = \sum_{i=0}^{N} \frac{N!}{i!(N-i)!} q^{i} (1-q)^{N-i} \max\left[u^{i} d^{N-i} Fl - Fx, 0\right]$$
(7)

Note that there is no explicit discounting in the formula. The discounting is implicit, since Fl and Fx are already present values This is how the original CRR model is applied to price European swaptions. In what follows, we will change the probabilities from qto $\frac{1}{2}$ in order to make the computations more efficient. This change of probability also affects both the up and the down factors. The details can be found in Jarrow and Rudd (1983).

2 The Binomial Model for Bermudan Swaptions

The object of analysis here is a Bermudan swaption with exercise dates $T_1, T_2, ..., T_N$. We consider a time interval $0 \leq t \leq T_N$. At this interval, there exists a collection of European swaptions O_{t,T_i} and spot swaps S_{t,T_i} with maturity dates T_i . The economic setting is a frictionless financial market where a given set of swaps and European swaptions are traded on a given finite time interval (all time points mentioned below are assumed to fall within this interval). The interval between trades is of length $\Delta > 0$, measured in units of time. The total number of steps until the final maturity date is n and steps to each T_i (maturity date), is given by T_i/Δ .

In order to construct a feasible and efficient binomial tree, we set the probabilities of up and down movements equal to 1/2. The continuous-time dynamics of the forward swap rates are given by (4) with the same Wiener process driving all the forward swaps, but with separate volatility parameters. There exists a zero coupon bond for each T_i , denoted by P_{t,T_i} . We start by approximating the dynamics of the forward floating leg processes using the discretization

$$Fl_{T_i,T_s,T_s} = Fl_{t,T_i,T_s} \exp\left(-\sum_{k=1}^{T_i/\Delta} \frac{\sigma_{T_i}^2}{2} \Delta + \sum_{k=1}^{T_i/\Delta} \sigma_{T_i} \rho_k \sqrt{\Delta}\right)$$
(8)

where T_S denotes the time of maturity of the swap underlying the option. The random variables ρ can take values in 1 and -1 with probability 1/2 for each realization. The random variables are independent and identically distributed. In addition, they are the same for all forward floating legs. In the limit, the discrete process (8) converges to its continuous-time counterpart

$$Fl_{T_i,T_s,T_s} = Fl_{t,T_i,T_s} \exp\left(-\frac{\sigma_{T_i}^2}{2}T_i + \sigma_{T_i}W_{T_i}\right)$$
(9)

The volatility parameters in (8) and (9) can be obtained directly from the volatility matrix for European swaptions. Hence, we construct a binomial tree for each floating leg, Fl_{T_i} , using the market volatilities in equation (8). The initial, time 0, values for each Fl_{T_i} are calculated directly from the discount curve, using equation (1).

With the given set up, we can now proceed by building a binomial tree for Fl_{T_N} , i.e. the floating leg underlying the longest maturity European swaption. This represents the last exercise opportunity of a Bermudan swaption and therefore gives the end-node values for the binomial tree. To simplify notation somewhat, we drop the subscripts denoting the maturity of the floating and fixed legs. It should be clear that for each floating and fixed leg, the maturity is equal to the last exercise opportunity of the Bermudan swaption plus the remaining term. Now, the final node values are given by 4

$$\max\left[Fl_{T_N}^j - Fx_{T_N}, 0\right] \tag{10}$$

with $Fl_{T_N}^j$'s obtained from equation (8). Fx_{T_N} is the exercise price, taken from the initial discount curve. After the end node values have been calculated for each state j, using (10), the recursive valuation methodology can be applied. From the last exercise opportunity, we move on to consider exercising at date T_{N-1} . As the exercise opportunities of Bermudan swaptions occur usually annually or semi-annually, we do not have to consider each step, as in the American option case. Instead, it is enough to step straight to the time point where there is potentially an early exercise opportunity. The value of a Bermudan swaption at the second last exercise opportunity is given by

$$\max\left[Fl_{T_{N-1}}^{j} - Fx_{T_{N-1}}, \mathbf{E}_{T_{N-1}}^{j}\left[\max\left[Fl_{T_{N}} - Fx_{T_{N}}, 0\right]\right]\right]$$
(11)

where the expectation is a conditional expectation evaluated at each particular node j. The graph below illustrates this

EXHIBIT 2 Calculation of conditional expectations



The optimality of early exercise is checked using (11). The next step is to make the

analogous calculations for the preceding steps, the general formula for this is

$$\max\left[Fl_{T_{i-1}}^{j} - Fx_{T_{i-1}}, BO_{T_{i}}^{j}\right]$$
(12)

where $BO_{T_i}^j$ is the expected value, at time T_{i-1} of the Bermudan swaption at time T_i , i.e. at the next exercise opportunity. The expected value is calculated for every state of nature j. Therefore, we calculate the value of the Bermudan swaption using iterated conditional expectations so that the probabilities are set to 1/2. Moreover, the binomial distributions of the underlying floating leg values are generated using the market quoted volatilities (potentially different for each underlying floating leg) and the initial values of the floating legs are extracted directly from the discount curve. To find the present value of the Bermudan swaption, we finally calculate the expected value today of the Bermudan swaption at the first exercise opportunity, T_1 .

2.1 Market Data

To implement the binomial pricing algorithm for pricing Bermudan swaptions, we first need to supply the necessary market data. This data includes the par swap curve and the volatility matrix for European swaptions. In order to calculate the initial values of the floating leg and fixed leg (strikes) values, a bootstrapping method need to be used to find the discount bond prices. After that, the required variables are calculated using equations (1) and (2).

The volatility matrix for European swaptions gives the Black (1976) swaption volatilities for at-the-money swaptions. On the other axis are the maturities of the underlying swaps, whereas the other axis gives the swaption maturities. Therefore, in order to price a ten-year Bermudan swaption using the binomial model presented here, we need to read specific entries from the volatility matrix. Of course, the entries in this case are 1y/9y, 2y/8y, 3y/7y,...,9y/1y. Here the first and second numbers represent the swaption maturity and the underlying swap maturity, respectively. These volatilities are the volatilities of the European swaptions maturing at the time of exercise opportunities of the Bermudan swaption. Each European swaption has the same term for the underlying swap as the Bermudan has, should it be exercised.

Working with the volatility matrix presents two difficulties in practice. The first is a mismatch of the tenors of the European swaptions and the Bermudan swaption that should be valued. This problem is easily overcome by using a two-dimensional interpolation scheme. The second problem is due to the volatility matrix data being quoted for at-the-money European swaptions. Almost always, the Bermudan swaption has a constant strike yield. Therefore, at-the-money European swaption data is not fully appropriate for valuation. A reasonable remedy for this would be to obtain smile data directly from brokers, at least for the most valuable European swaptions.

2.2 Variable Strike Bermudan

Using the binomial algorithm in pricing a Bermudan swaption with a variable strike is straightforward. As the strikes in the OTC interest rate option market are quoted as rates, we only need to apply these strike rates in the calculation of fixed leg values for individual exercise opportunities. Obviously, if the strike rates deviate significantly from the quoted at-the-money rates, then the volatility matrix figures need to be adjusted.

2.3 Hedging Parameters

Derivation of the hedging parameters is similar to the standard binomial model. However, in this case, there is a freedom in the choice of the hedging instrument, i.e. which of the underlying forward swaps to use. From the available choices, there are two that stand out. The first is to use the forward swap underlying the most valuable European swaption. Since the value of that instrument accounts for a large part of the value of the Bermudan swaption in total, it is likely that the hedging error is not going to be exceedingly large if that particular forward swap is used. The second good alternative is to use some combination of the underlying forward swaps. This reduces the effect of the yield curve risk in case of non-parallel movements. Conceptually this is similar to bucket management of risk. Analysis and management of risk based on bucketing is a widely-used approach in practice.

The formula for calculation of the Delta is

$$Delta_t \frac{(BO_t^{up} - BO_t^{down})}{(u_{T_i} - d_{T_i})F_{t,T_i}}$$
(13)

where

$$u_{T_i} = e^{-\frac{\sigma_{T_i}^2}{2}\Delta + \sigma_{T_i}\sqrt{\Delta}}$$

$$d_{T_i} = e^{-\frac{\sigma_{T_i}^2}{2}\Delta - \sigma_{T_i}\sqrt{\Delta}}$$
(14)

3 Numerical Examples

We illustrate the use of the binomial model by applying it to the pricing of a Bermudan interest rate swaption. In addition, we compare the prices given by the binomial model to BDT model values.

Further assume that we want to price a Bermudan payer swaption with the maturity of two years, exercisable after one or two years. Suppose that the initial data values are as follows and the strike rate is 6.00. As indicated previously, the probabilities in the tree are 1/2 throughout.

The data: N = 2, K=1, D=0.5, and

From the initial swap market data, we calculate the values of the floating and the fixed legs (strike prices) and the corresponding value trees using the dynamics given by (8). The resulting trees with end-node option values are shown in Exhibit 3 below

EXHIBIT 3 Binomial Evolution of Floating Legs and End Node Values of a Payer Swaption with Strike 6.00%



Using the formula (7), the current price of the European payer swaption with time to maturity of two years and the underlying swap length of one year is

$$EO_t = 0.5^4 * (0.031 + 4 * 0.016 + 6 * 0.003)$$
$$= 0.0071$$

To price a Bermudan swaption, we need to check the possibility for early exercise. We start by observing the intrinsic values at the end nodes of the 2y/1y tree. Using these values we calculate the prices of Bermudan options at end-nodes of the 1/2 tree (the first exercise opportunity). According to equation (11), we find that in the upper node,

the value of the Bermudan swaption with strike rate 6.00% is

$$\max\left[0.1304 - 0.1044, 0.25 \times 0.0317 + 0.5 \times 0.0159 + 0.25 \times 0.0032\right] = 0.0260$$

Hence, early exercise is optimal. Similar calculations give: 0.0056 (no early exercise) for the middle-node and 0.0008 (no early exercise) for the lower-node. Finally, we can calculate the present value of the Bermudan swaption as follows

$$BO_t = (0.25 \times 0.0260 + 0.5 \times 0.0056 + 0.25 \times 0.0008) = 0.0095$$

In order to calculate the delta we need the up and down values for the Bermudan swaption. In addition, assume that we are going to implement the hedging with the two-year swap starting after one year. The Delta parameter in this case becomes

$$Delta_t = \frac{(0.0158 - 0.0032)}{(1.1009 - 0.9160) \times 0.1076} = 0.6332$$

The procedure described here illustrates how the Bermudan swaption values can be calculated using a very simple binomial algorithm. In practise, the number of steps between the exercise opportunities would usually be between 100 and 1000. The algorithm is very quick, so this does not present any problem.

To gain some insight on how the binomial model values Bermudan swaptions, we compare the simple binomial model with the BDT model that is widely used by market participants. We fitted the BDT model to market data using a tree with 100 steps. Three different yield curve environments were used: flat, linear upward and linear downward. The volatility structure was flat in all cases, but the level varied from 10 to 20 percent. In calculating the option values, we set the number of steps between exercise opportunities to 100. The results are shown in Exhibit 4 below

Tenor	BDT Vol	Black Vol	Curve	Calibration	BDT Price	Binomial
10 yr	14.90%	15.0%	Flat 5.00%	Atm fwd	352.19	346.70
10 yr	15.02%	15.0%	Up 4.00%-6.00%	Atm fwd	812.09	804.83
10 yr	14.79%	15.0%	Dn 6.00%-4.00%	Atm fwd	59.65	57.84
10 yr	9.87%	10.0%	Flat 5.00%	Atm fwd	234.62	231.66
10 yr	9.91%	10.0%	Up 4.00%-6.00%	Atm fwd	711.68	709.84
10 yr	9.82%	10.0%	Dn 6.00%-4.00%	Atm fwd	11.67	11.39
10 yr	20.12%	20.0%	Flat 5.00%	Atm fwd	471.03	461.06
10 yr	20.36%	20.0%	Up 4.00%-6.00%	Atm fwd	927.55	911.66
10 yr	19.89%	20.0%	Dn 6.00%-4.00%	Atm fwd	132.59	127.44
5 yr	14.69%	15.0%	Flat 5.00%	Atm fwd	135.43	134.65
5 yr	14.69%	15.0%	Up 4.00%-6.00%	Atm fwd	172.41	172.37
5 yr	14.68%	15.0%	Dn 6.00%-4.00%	Atm fwd	106.96	106.53
5 yr	9.79%	10.0%	Flat 5.00%	Atm fwd	90.43	89.85
5 yr	9.79%	10.0%	Up 4.00%-6.00%	Atm fwd	129.20	128.98
5 yr	9.78%	10.0%	Dn 6.00%-4.00%	Atm fwd	64.33	64.56
5 yr	19.62%	20.0%	Flat 5.00%	Atm fwd	180.51	179.24
5 yr	19.64%	20.0%	Up 4.00%-6.00%	Atm fwd	217.35	216.74
5 yr	19.61%	20.0%	Dn 6.00%-4.00%	Atm fwd	150.11	148.96

EXHIBIT 4 Simple Binomial Algorithm and BDT Model Compared

Exhibit 4 shows that, in general, the prices given by the simple binomial model and the BDT model are fairly similar. The shorter maturity swaption prices, in particular, are almost equal. With longer maturity swaptions, the divergence is more pronounced, increasing with the level of volatility. This may be due to the fact that the valuation tree is relatively scarcely sliced. Given that the binomial algorithm is far simpler to implement, calibrates automatically, doesn't need an iterative algorithm to solve for node values, it provides a good alternative to the traditional interest rate models for valuation of Bermudan type swaptions. In particular, if the aim is a quick and reliable valuation using market data, then the best method is to implement the simple binomial algorithm presented in this paper.
4 Application to Commodity

It is possible to extend this simple binomial model to handle commodity Bermudan swaptions as well. Although the model is overly simplistic in the assumed dynamics (i.e. one driving Brownian motion for the set of forward par swap prices, scaled by the individual constant volatility parameters) it is still an applicable method in the presence of suitable market data. The model implemented here closely resembles the ideologies put forth in the papers by Reisman (1991) and Cortazar and Schwartz (1994). That is, instead of modelling forward prices through the stochastic model for the spot price and net convenience yield, the dynamics of the prices is exogenously given. In that sense, the effect of a stochastic net convenience yield, which is able to generate mean-reversion to the prices and affect the volatility structure is ignored. It is fair to say that in the absence of reliable market data on both the forward prices and forward price volatilities, the model is not applicable in practice. However, if one does have good data to use and a liquid market to trade in, then even this simple one-factor implementation is able to produce useful results. A recent paper by Driessen, Klaassen, and Melenberg (2003) shows that the hedging performance of one-factor versus multi-factor models is approximately equal if one uses a many underlying instruments i.e. bucket hedging, which is a common practice among trading institutions. Even though they analyzed data on US interest rate cap and swaption prices, the general observation of the disappearing advantage of multi-factor models when implementing bucket hedging strategies is likely to be valid when applied to other markets and instruments as well.

In order to describe how the model is to be adjusted for handling commodity Bermudan swaptions, we need to redefine the underlying variables: the value of the floating leg and the value of the fixed leg. The present value of the floating price leg at time t is given by

$$Fl_{t} = \sum_{i=1}^{N} F(t, T_{i}) P(t, T_{i})$$
(15)

where $F(t, T_i)$'s are the implied forward prices from the forward curve. See Järvinen (2002) for various methods to handle the derivation of the forward curve from swap quotes. In commodity implementation, the value of the floating leg has to be calculated forward price by forward price and the sum of the discounted forward prices finally gives the present value of the floating leg of the swap. The value of the fixed leg at time t is given by

$$Fx_t = X \sum_{i=1}^{N} P(t, T_i)$$
 (16)

Note that there are no day count conventions used, since the fixings of a commodity swap are defined as notional multiplied by the fixed price, X. Multiplying these fixing values by the appropriate discount factors gives the present value of the fixed leg.

Analogously with the analysis of the interest rate instruments, by setting the values of the floating price and fixed price equal we can solve for the par forward swap price, i.e. the forward swap that makes the present value of the contract zero. Solving for Xgives the par swap price S_t

$$S_{t} = \frac{\sum_{i=1}^{N} F(t, T_{i}) P(t, T_{i})}{\sum_{i=1}^{N} P(t, T_{i})}$$
(17)

Moreover, we assume that the par swap price follows geometric Brownian motion with the drift rate of zero, i.e. a gaussian martingale process for the log par swap price. Hence, the European commodity swaption prices can be calculated by using the standard Black (1976) formula, i.e. equation (5). Järvinen and Toivonen (2002) presents detailed arguments for arriving at this result.

After the underlying variables have been calculated to enable the building of the

multilayer binomial tree, the rest of the analysis is analogous to the interest rate case. In contrast to the interest rate market, where data on European swaption volatilities on both maturity and strike axis is available, the data from the commodity derivatives market is definitely more difficult, if not impossible, to obtain. Therefore, the practical applicability of this pricing methodology is suspect, or at least varies across commodity types.

5 Conclusion

A simple binomial algorithm enables the calculation of Bermudan swaption values and hedges very efficiently. More importantly, it enables inclusion of the market volatilities and forward swap rate information instantly, without the need for any sort of calibration. The algorithm itself is based on the standard binomial model, with probabilities set to 1/2. The underlying assumption is that there is only one source of randomness that drives the values of all the forward swaps (or forward floating legs) simultaneously, each weighted by their own Black (1976) market volatilities. Discounting is assumed constant.

It is demonstrated that the model prices do not differ much from the Bermudan swaption prices given by the BDT model fitted to the yield curve. This suggests that, in particular for those market parties who need a good and fast estimate of the value of Bermudan swaptions or multi-call bonds, the algorithm provides a cost efficient valuation method. Also traders, experienced with the Black (1976) model for European caps and swaptions, will find the simple binomial algorithm a useful tool for the analysis of Bermudan swaptions.

Notes

¹Recently, many feasible methods for adapting the Monte Carlo method to handle early-exercise problems have been proposed. One of the major motivations for these studies have been to find ways of implementing the multifactor Libor Market Models, LMM's (Brace, Gatarek, and Musiela (1997) and Miltersen, Sandmann, and Sondermann (1997)) to price also Bermudan swaptions.

²Bjerksund and Stensland (1996) have proposed an analytical approximation.

³There exists various methods for calculating interest rate day count factors, τ . Some of the most often used are act/360, act/act and 30/360, where act is the actual number of days applying to the interest accrual period.

⁴For a receiver option, this becomes $\max \left[Fx_{T_N} - Fl_{T_N}^j, 0\right]$

References

- Andersen, L., 1999, "A Simple Approach to the Pricing of Bermudan Swaptions in the Multi-Factor Libor Market Model," Working paper, Social Science Research Network.
- Barraquand, J., and D. Martineau, 1995, "Numerical Valuation of High Dimensional Multivariate American Securities," *Journal of Financial and Quantitative Analysis*, 30, 388–405.
- Bjerksund, P., and G. Stensland, 1996, "Implementation of the Black-Derman-Toy Interest Rate Model," *Journal of Fixed Income*, 6, 67–75.
- Black, F., 1976, "The Pricing of Commodity Contracts," Journal of Financial Economics, 3, 167–179.
- Black, F., E. Derman, and W. Toy, 1990, "A One-factor Model of Interest Rates and its Application to Treasury Bond Options," *Financial Analysts Journal*, 46, 33–39.
- Brace, A., M. Gatarek, and M. Musiela, 1997, "The Market Model of Interest Rate Dynamics," *Mathematical Finance*, 7, 127–155.
- Broadie, M., and P. Glasserman, 1997, "A Stochastic Mesh Method for Pricing High-Dimensional American Options," Working paper, Columbia University, New York.
- Carr, P., and G. Yang, 1997, "Simulating Bermudan Interest Rate Derivatives," Working paper, Courant Institute at New York University.
- Cortazar, G., and E. S. Schwartz, 1994, "The Valuation of Commodity Contingent Claims," *Journal of Derivatives*, 1, 27–39.
- Cox, J., S. Ross, and M. Rubinstein, 1979, "Option Pricing: A Simplified Approach," Journal of Financial Economics, 7, 229–263.

- Driessen, J., P. Klaassen, and B. Melenberg, 2003, "The Performance of Multi-Factor Term Structure Models for Pricing and Hedging Caps and Swaptions," *Journal of Financial and Quantitative Analysis*, 38, 635–672.
- Hull, J., and A. White, 1990, "Pricing Interest Rate Derivative Securities," Review of Financial Studies, 3, 573–592.
- Jamshidian, F., 1997, "LIBOR and Swap Market Models and Measures," Finance and Stochastics, 1, 293–330.
- Jarrow, R., and A. Rudd, 1983, Option Pricing. Richard D. Irwin, Inc., Illinois, US.
- Järvinen, S., 2002, "Estimating the Forward Curve for Commodities," Unpublished manuscript, Helsinki School of Economics.
- Järvinen, S., and H. Toivonen, 2002, "Pricing European Commodity Swaptions," Unpublished manuscript, Helsinki School of Economics.
- Longstaff, F., and E. Schwartz, 2001, "Valuing American Options by Simulation: A Simple Least-Squares Approach," *Review of Financial Studies*, 14, 113–147.
- Miltersen, K., K. Sandmann, and D. Sondermann, 1997, "Closed-form Solutions for Term Structure Derivatives with Lognormal Interest Rates," *Journal of Finance*, 52, 409–430.
- Pedersen, M., 1999, "Bermudan Swaptions in the LIBOR Market Model," Working paper, Social Science Research Network.
- Rebonato, R., 1996, Interest-rate Option Models. John Wiley and Sons, Chichester, England, 1 edn.
- Reisman, H., 1991, "Movements of the Term Structure of Commodity Futures and the Pricing of Commodity Claims," Unpublished manuscript, Haifa University, Israel.

ESSAY 4:

Pricing European Commodity Swaptions

Sami Järvinen and Harri Toivonen

Abstract

In this paper, we present formulas for commodity swaptions. By utilizing the forward price based approach we derive a simple closed form solution for European swaptions based on the assumption of deterministic volatility for lognormal variables. The formulas given result from applying the Margrabe (1978) exchange option concept to the present problem. A special case of constant volatility yields the Black (1976) formula that has been the market standard in the interest rate swaption markets for many years. Futures markets have been the traditional vehicle for participating in the commodities markets, and exchange-traded futures contracts on commodities have a long history dating back to the 1800s. In fact, the first standardized derivative contracts were traded on commodities. In recent years, the OTC market for commodity derivatives has also expanded rapidly: BIS (2002) reports that the notional amount outstanding of OTC commodity derivative contracts was USD 590 billion in June 2001¹. The increasing financial importance of these products has also led to growing interest among academic researchers.

Modelling the stochastic behavior of underlying commodity prices is important for the valuation of contingent claims on commodities. Earlier studies on commodity derivatives have concentrated on modelling the joint stochastic process for net convenience yield and spot commodity price; a seminal paper along these lines is Gibson and Schwartz (1990)² who develop a two-factor model where the state variables are spot price and mean-reverting instantaneous net convenience yield. Schwartz (1997) extends this approach by introducing three alternative stochastic models, where the most sophisticated has three state variables: spot price of commodity, convenience yield and interest rate. Miltersen and Schwartz (1998) present a different approach and extend earlier models by utilizing information in the initial term structure of interest rates and futures prices, thereby obtaining a more market-oriented model in the spirit of the classic Heath, Jarrow, and Morton (1992) paper. In addition, by assuming normal distribution for instantaneous interest rates and convenience yields, Miltersen and Schwartz (1998) are able to derive simple closed form solutions for futures and forward prices. Hilliard and Reis (1998) present a three-factor model which allows for jumps in the spot price. In addition, they develop an equilibrium-based diffusion for the convenience yield process and an arbitrage free term structure process. Hilliard and Reis (1998) make this choice for applicability; convenience yield data is hard to obtain, but interest rate data is reliably available.

Hence, in practice, in order to price commodity derivatives, it is often more convenient to model the evolution of the forward variables instead of explicitly modelling the spot variables, especially the rather abstract convenience yield. This approach was pioneered by Black (1976). He priced options directly in terms of the forward price of the underlying asset with general stochastic dividends, though he kept the interest rates constant. Since then, the forward approach to arbitrage valuation of contingent claims has gained popularity. Bick (1998) developed a general forward price based valuation framework and replication strategy. More recently, the numeraire change based technique, formalized by Geman, Karoui, and Rochet (1995), has become the standard tool for derivatives valuation, and it has been used to give mathematically formal justifications for many important pricing models, e.g. the Black formula for interest rate swaptions, originally developed by Neuberger (1990) and later formalized by Jamshidian (1997).

Chance and Rich (1996) have previously introduced a methodology for pricing commodity swaptions. However, in order to use their pricing formula in practice, one needs to know the continuously compounded convenience yield. In this paper, we present a valuation formula for a European commodity swaption, based on the forward price approach, i.e. we specify the dynamics of the forward price exogenously. Within this framework, no information regarding the stochastic process of the net convenience yield is required in order to price commodity instruments. This approach simplifies the pricing of commodity swaptions considerably. To the best of our knowledge, to date, there have been no published studies where the forward based approach has been applied to the pricing of commodity swaptions. We contribute to the existing literature by presenting a general framework for pricing commodity swaptions using the change of numéraire technique by Geman, Karoui, and Rochet (1995). Use of the forward price process is arguably a more market-oriented approach than using the joint processes for spot variables. The results presented in this paper also include the effect on the pricing formula if one allows for stochastic interest rates.

The paper is organized as follows: An exact definition of a commodity swaption is given in Section 1. The pricing formula for both the constant and stochastic interest rates are derived in Section 2. Extensions are presented in Section 3. Finally, Section 4 discusses the validity of the approach and Section 5 concludes the paper.

1 Basic Definitions

A commodity swaption is an option that gives the holder a right to pay (payer swaption) or receive (receiver swaption) a fixed price against the floating price of the underlying commodity. In many ways, the commodity swaption is similar to the standard interest rate swaption. However, there are some important differences: First, the notional amount of the commodity swaption is in tonnes or barrels or other units of the underlying commodity, whereas the notional amount of the interest rate swaption is a currency amount. In addition, the settlement prices of the underlying swap are often calculated from multiple price observations instead of a single observation that is typically the case with interest rate swaps. The underlying instrument in the commodity swaption, when it matures, is a spot commodity swap. However, it is often more convenient to cast the actual pricing of the contract into the framework where forward start commodity swap is used as the underlying instrument. Let us fix a collection of future dates $T_0 = T < T_1 < \ldots < T_N$. The forward swap is a financial contract entered into at trade date $t < T_0$ with settlement dates $T_1 < ... < T_N$. Often the spot price of the commodity or a proxy index, is observed daily or weekly between the settlement dates and that value is then used as the settlement price. At each settlement date of the swap, the other party pays the difference $S(T_i) - X$, where $S(T_i)$ is the floating price for settlement T_i and X is the agreed fixed price.

A forward swap contract is composed of two legs, namely the floating price leg and the fixed price leg. The present value of the floating price leg is given by

$$V_{float} = \sum_{i=1}^{N} F(t, T_i) P(t, T_i)$$

$$\tag{1}$$

i.e. it is the sum of the discounted forward prices. Usually, the swap contracts exist for maturities far exceeding the longest dated available forward or futures contracts. By contrast, the spot swap prices are quoted for maturities of many years, typically at least five years. Therefore, the forward prices in (1) have to be implied from the swap market data using a suitable estimation algorithm - see Järvinen (2002) ³.

The present value of the fixed price leg can be obtained by the following formula

$$V_{fixed} = X \sum_{i=1}^{N} P(t, T_i)$$
⁽²⁾

So, the fixed price leg value is simply the sum of the discount factors for the settlement dates of the swap contract multiplied by the fixed price of the swap. The sum of discount factors is often called annuity. Both of the leg values are multiplied by notional per fixing to find the actual currency denominated values. In equations (1) and (2), the notional quantity is assumed to be one to simplify notation.

By setting the values of the floating price and fixed price leg values equal, we can solve for the par forward swap price, i.e. the forward swap which makes the present value of the contract zero. Solving for X gives

$$X = \frac{\sum_{i=1}^{N} F(t, T_i) P(t, T_i)}{\sum_{i=1}^{N} P(t, T_i)}$$
(3)

The payoff of the European commodity swaption can be defined as the maximum of the

values of the floating price leg and the fixed price leg. We have for the payer swaption

$$C(T_0) = \max\left[V_{float}(T_0) - V_{fixed}(T_0), 0\right]$$
(4)

and for the receiver swaption

$$P(T_0) = \max\left[V_{fixed}(T_0) - V_{float}(T_0), 0\right]$$
(5)

In effect, the payer swaption is a call on the floating price leg, and the receiver swaption is a put on the floating price leg. From the homogeneity of the payoff functions (4) and (5), it follows that we can rewrite the payoffs as

$$C(T_0) = \sum_{i=1}^{N} P(T_0, T_i) \times \max\left[S(T_0) - X, 0\right]$$
(6)

$$P(T_0) = \sum_{i=1}^{N} P(T_0, T_i) \times \max\left[X - S(T_0), 0\right]$$
(7)

In other words, the swaption payoffs can be rephrased as the annuitized difference between the spot swap price at date T_0 and the fixed strike price agreed upon the contract initiation. The rewritten payoff functions can be compared with the interest rate swaption payoffs. First, the spot swap price corresponds to the spot swap rate in the interest rate swaption with the fixed strike price having an analogous interpretation. Second, the price difference is annuitized, which corresponds to present valuing the rate difference in the interest rate swaption. Third, there are no day counts or annual notional amount. Instead, there is a notional amount per fixing.⁴ In the case of commodity swaps and swaptions, the notional amount is not a monetary unit, but rather it is the quantity of the underlying commodity's unit of quotation.

2 The Pricing Formulas

Valuing a commodity swaption requires the specification of the dynamics for the underlying variables, namely, the forward floating price leg value and the forward fixed price leg value.

We assume that the dynamics under the real-world probability \mathbb{P} are given by

$$dV_{float}(t) = \alpha_1(t)V_{float}(t)dt + \sigma_1(t)V_{float}(t)dW_t^1$$
(8)

and

$$dV_{fixed}(t) = \alpha_2(t)V_{fixed}(t)dt + \sigma_2(t)V_{fixed}(t)dW_t^2$$
(9)

where dW_t is the standard Brownian motion under \mathbb{P} and $\alpha_{1,2}(t)$ and $\sigma_{1,2}(t)$ are drift and volatility coefficients, respectively. In addition, the standard Brownian motions are correlated with instantaneous correlation given by $\rho(t)dt$. As in Bick (1998) we assume no specific processes for convenience yield and interest rates. Instead, the forward processes (8) and (9) are modelled as correlated geometric Brownian motions. This facilitates straightforward calculation of swaption values. Under the traditional riskneutral measure \mathbb{Q} , the drift coefficients of the processes (8) and (9) become r(t), which denotes the continuously compounded risk-free rate of return.⁵

In order to derive simple formulas, we model the forward floating leg and the forward fixed leg values by summing up all the associated elements $P(t, T_i)F(t, T_i)$ and $P(t, T_i)$, respectively. This way, assuming that the volatility of the forward swap price is deterministic, it then follows the lognormal law. In analyzing (6) and (7) we are effectively calculating the expected value of the option payoff using the forward fixed price leg divided by the strike price as the numéraire (this is also known as annuity). This is an application of the general numéraire change technique developed by Geman, Karoui, and Rochet (1995). Our results are direct applications of the general theorems established in their paper. We get the following expression for the expectation

$$\frac{V_{float}(0)}{A_{N,T}(0)} = \mathbb{E}^A \left[\frac{V_{float}(t)}{A_{T_1,T_N}(t)} \right]$$
(10)

where \mathbb{E}^A denotes expectations under the annuity measure and $A_{T_1,T_N}(t) = \sum_{i=1}^N P(t,T_i)$. It is well known that under this change of numéraire, the forward swap price is a martingale and the expected spot swap price equals the forward swap price. In fact, given our assumptions, the problem of finding the value function for European commodity swaption is analogous to exchange option problem, introduced by Margrabe (1978). The strike price X can be interpreted as simply a scaling factor. The equation (10) generalizes to the contingent claims valuation function, where f denotes any attainable simple claim, having T_0 as maturity date

$$f(0) = A_{T_1, T_N}(0) \times \mathbb{E}^A \left[\frac{f(T_0)}{A_{T_1, T_N}(T_0)} \right]$$
(11)

Substituting the payer commodity swaption pricing formula in (11), we get

$$C(0) = A_{T_1,T_N}(0) \times \mathbb{E}^A \left[\frac{A_{T_1,T_N}(T_0) \times \max[S(T_0) - X, 0]}{A_{T_1,T_N}(T_0)} \right]$$

= $A_{T_1,T_N}(0) \times \mathbb{E}^A \left[\max[S(T_0) - X, 0] \right]$

This is the payoff of the standard European call. From the assumption of deterministic volatility of the forward swap price, we have the familiar results

$$C(t) = \sum_{i=1}^{N} P(t, T_i) \left[S(t, T_0, T_N) N(d_1) - X N(d_2) \right]$$
(12)

and

$$P(t) = \sum_{i=1}^{N} P(t, T_i) \left[XN(-d_2) - S(t, T_0, T_N)N(-d_1) \right]$$
(13)

where

$$d_{1} = \frac{\ln (S(t, T_{0}, T_{N})/X) + \frac{1}{2} \int_{t}^{T_{0}} \|\sigma_{1}(u) - \sigma_{2}(u)\|^{2} du}{\sqrt{\int_{t}^{T_{0}} \|\sigma_{1}(u) - \sigma_{2}(u)\|^{2} du}}$$

$$d_{2} = d_{1} - \sqrt{\int_{t}^{T_{0}} \|\sigma_{1}(u) - \sigma_{2}(u)\|^{2} du}$$

where $\|\cdot\|$ denotes Euclidean norm. Equations (12) and (13) are solutions to the call (payer) and put (receiver) prices for commodity swaptions under our restrictive assumptions. Similar results have been derived already by Merton (1973) for the case of equity options with stochastic interest rates. In case of constant volatility and correlation coefficients, we will have

$$\sqrt{\int_{t}^{T_{0}} \|\sigma_{1}(u) - \sigma_{2}(u)\|^{2} du} = \sqrt{(\sigma_{1}^{2} + \sigma_{2}^{2} - 2\rho\sigma_{1}\sigma_{2})(T_{0} - t)}$$
(14)

and we have a version of the famous Margrabe (1978) formula for options to exchange one asset for another. Furthermore, if we set $\sigma_2(t) = 0$, then the Black (1976) formula is recovered. To interpret the formula, we note that if the stochastic interest rate environment is assumed, then the implied volatility of the asset is actually the implied volatility of the forward swap price. Only in the constant interest rate world is the interpretation of volatility being the implied volatility of the asset correct.

3 Extensions

Due to the simplicity of the assumed dynamics, it is straightforward to extend the model to include more driving factors providing one makes sure that the forward swap price has a deterministic volatility. Moreover, the model can easily be extended to price structures such as swaptions on swaps with spread, cross currency swaptions, quanto swaptions and many popular exotic swaptions. These results have already been established for the standard options assuming the Black and Scholes (1973) market. We give examples of pricing a swaption with spread and an amortizing notional swaption. A swaption on a swap with spread is similar the same as the swaption apart from the fact that there is a spread attached to the floating price leg of the swap underlying the swaption. As the spread is a fixed value, we therefore adjust the strike price to account for the spread and treat the floating price leg as a pure floating price leg. Hence, the spread L, is subtracted from the fixed price leg value

$$V_{fixed-adjusted} = (X - L) \sum_{i=1}^{N} P(t, T_i)$$
(15)

Now, the swaption payoff becomes

$$C_{T_0} = \max\left[V_{float}(T_0) - V_{fixed-adjusted}(T_0), 0\right]$$
(16)

and again, from linear homogeneity

$$C_{T_0} = \sum_{i=1}^{N} P(T_0, T_i) \times \max\left[S(T_0) - (X - L), 0\right]$$
(17)

Thus, the swaption with a spread can now be valued in the same way as the ordinary swaption with strike price replaced by (X - L).

Amortizing notional can be handled as in Jamshidian (1997). Let us denote the

notional quantity of the underlying swap on settlement date T_i by $L(T_i)$. Then equations (1) and (2) become

$$V_{float} = \sum_{i=1}^{N} L(T_i) F(t, T_i) P(t, T_i)$$
(18)

and

$$V_{fixed} = X \sum_{i=1}^{N} L(T_i) P(t, T_i)$$
 (19)

Now, the "break-even" forward swap price will be

$$X = \frac{\sum_{i=1}^{N} L(T_i) F(t, T_i) P(t, T_i)}{\sum_{i=1}^{N} L(T_i) P(t, T_i)}$$
(20)

and again, assuming that this "break-even" price has deterministic volatility, the results given earlier apply. As noted by Jamshidian (1997), the assumptions of deterministic volatility for different swap prices may not be consistent with each other, but individually they appear to be quite reasonable, robust and convenient.

4 Discussion

The standard approach to analyzing commodity derivatives has traditionally been based on stochastic models of the spot price, convenience yield and interest rates. The approach presented here relies on modelling the forward prices directly without making any statements on the stochastic behavior of convenience yield. Moreover, in order to establish a simple formula for pricing swaptions, we treat the ratio of the forward floating leg value and the forward fixed leg value as a stochastic variable with deterministic volatility. That is, instead of modelling the underlying forward swap price as a sum of weighted individual forward price dynamics, we first put the package together and only then define the stochastic model. This approach creates a modelling problem in the sense that if the forward swap price follows the lognormal distribution, then the components (forward prices) can not be lognormal simultaneously. This problem is already known from the interest rate derivatives modelling, where the Libor market models (Miltersen, Sandmann, and Sondermann (1997)) and (Brace, Gatarek, and Musiela (1997)) cannot be valid simultaneously with the Swap market model (Jamshidian (1997)). Even Libor market models configured for certain tenor structure, i.e. quarterly, semi-annual, etc., are inconsistent with each other. We refer to Jamshidian (1997) for an excellent discussion of the merits of using different models for different products despite the apparent inconsistency between the model's assumptions.

In summary, the market for commodity derivatives is still rather young compared to, for example the market for interest rate derivatives, where modelling approaches have evolved from simple instantaneous short rate models to the Libor market models currently being the tool of analysis in the sophisticated investment banks. In contrast, the commodity markets are much more heterogeneous in terms of the institutional settings and details (such as the liquidity and quotation practices) of the market for the underlying commodities, as well as in the products themselves. It remains to be seen what the standard market models for these instruments will be in the future. This paper attacks the problem from the point of view of pricing in terms of the forward variables, and the validity of such an approach obviously depends upon the depth of the market for forward contracts, swaps and plain vanilla options. If there is enough liquidity in those contracts, then there is certainly a good argument for applying the forward price based approach.

5 Conclusions

This paper presents a Black-76 formula for commodity swaptions. The underlying stochastic variables are the values of the floating and fixed price legs. Using the annuity factor as the numéraire, the familiar closed form solution for the swaption price is obtained. Interest rate market participants have long used the interest rate swaption formula, derived initially by Neuberger (1990), to price plain vanilla interest rate swaptions. For commodity swaptions, a similar formula is shown to exist, with the same distributional assumptions. Hence, it is very likely that the traders, familiar with the Black-76 volatilities, will use the model as a base when the OTC-market for commodity swaptions is large enough to support regularly quoted market prices.

Notes

¹The figure comprises of OTC contracts for commodity forwards, swaps and options.

²Convenience yield is defined as the benefit derived from holding the physical commodity instead of the futures contract on the underlying. Net convenience yield is convenience yield net of holding period storage costs.

³The quotations for longer maturity swaps are usually average-based, which complicates the bootstrapping process.

⁴The formula can be specified so that it uses annual notional and year fractions instead

⁵The forward floating price leg and the forward fixed price leg do not provide income before the start date of the spot swap, since convenience yields are already capitalized into the forward prices. This can be shown using Ito's Lemma.

References

- Bick, A., 1998, "Producing Derivative Assets with Forward Contracts," Journal of Financial and Quantitative Analysis, 23, 153–160.
- BIS, 2002, "BIS Quarterly Review March 2002," .
- Black, F., 1976, "The Pricing of Commodity Contracts," Journal of Financial Economics, 3, 167–179.
- Black, F., and M. Scholes, 1973, "The Pricing of Options and Corporate Liabilities," Journal of Political Economy, 81, 637–659.
- Brace, A., M. Gatarek, and M. Musiela, 1997, "The Market Model of Interest Rate Dynamics," *Mathematical Finance*, 7, 127–155.
- Chance, D., and D. Rich, 1996, "Asset Swaps with Asian-style Payoffs," Journal of Derivatives, 3, 64–77.
- Geman, H., N. E. Karoui, and J. Rochet, 1995, "Changes of Numeraire, Changes of Probability Measure and Option Pricing," *Journal of Applied Probability*, 32, 443– 458.
- Gibson, R., and E. Schwartz, 1990, "Stochastic Convenience Yield and the Pricing of Oil Contigent Claims," *Journal of Finance*, 45, 959–976.
- Heath, D., R. Jarrow, and A. Morton, 1992, "Bond Pricing and the Term Structure of Interest Rates: A New Methodology for Contingent Claims Valuation," *Econometrica*, 60(1), 77–105.
- Hilliard, J. E., and J. Reis, 1998, "Valuation of Commodity Futures and Options under Stochastic Convenience Yields, Interest Rates, and Jump Diffusions in the Spot," *Journal of Financial and Quantitative Analysis*, 33(1), 61–86.

- Jamshidian, F., 1997, "LIBOR and Swap Market Models and Measures," Finance and Stochastics, 1, 293–330.
- Järvinen, S., 2002, "Estimating the Forward Curve for Commodities," Unpublished manuscript, Helsinki School of Economics.
- Margrabe, W., 1978, "The Value of an Option to Exchange One Asset for Another," Journal of Finance, 33, 177–186.
- Merton, R. C., 1973, "The Theory of Rational Option Pricing," Bell Journal of Economics and Management Science, 4, 141–183.
- Miltersen, K., K. Sandmann, and D. Sondermann, 1997, "Closed-form Solutions for Term Structure Derivatives with Lognormal Interest Rates," *Journal of Finance*, 52, 409–430.
- Miltersen, K., and E. Schwartz, 1998, "Pricing of Options on Commodity Futures with Stochastic Term Structures of Convenience Yields and Interest Rates," *Journal of Financial and Quantitative Analysis*, 33, 33–59.
- Neuberger, A., 1990, "Pricing Swap Options Using The Forward Swap Market," London Business School Working Paper.
- Schwartz, E., 1997, "The Stochastic Behavior of Commodity Prices: Implications for Valuation and Hedging," *Journal of Finance*, 52, 923–973.

A-SARJA: VÄITÖSKIRJOJA - DOCTORAL DISSERTATIONS. ISSN 1237-556X.

- A:197. ANSSI ÖÖRNI: Consumer Search in Electronic Markets. 2002. ISBN 951-791-680-9.
- A:198. ARI MATIKKA: Measuring the Performance of Owner-Managed Firms: A systems approach. 2002. ISBN 951-791-685-X.
- A:199. RIITTA KOSONEN: Governance, the Local Regulation Process, and Enterprise Adaptation in Post-Socialism. The Case of Vyborg. 2002. ISBN 951-791-692-2.
- A:200. SUSANNE SUHONEN: Industry Evolution and Shakeout Mechanisms: The Case of the Internet Service Provider Industry. 2002. ISBN 951-791-693-0.
- A:201. MATTI TUOMINEN: Market-Driven Capabilities and Operational Performance. Theoretical Foundations and Managerial Practices. 2002. ISBN 95-791-694-9.
- A:202. JUSSI KARHUNEN: Essays on Tender Offers and Share Repurchases. 2002. ISBN 951-791-696-5.
- A:203. HENNAMARI MIKKOLA: Empirical Studies on Finnish Hospital Pricing Methods. 2002. ISBN 951-791-714-7.
- A:204. MIKA KORTELAINEN: EDGE: a Model of the Euro Area with Applications to Monetary Policy. 2002. ISBN 951-791-715-5.
- A:205. TOMI LAAMANEN: Essays on Technology Investments and Valuation. 2002. ISBN 951-791-716-3.
- A:206. MINNA SÖDERQVIST: Internationalisation and its Management at Higher-Education Institutions. Applying Conceptual, Content and Discourse Analysis. 2002. ISBN 951-791-718-X.
- A:207. TARJA PIETILÄINEN: Moninainen yrittäminen. Sukupuoli ja yrittäjänaisten toimintatila tietoteollisuudessa. 2002. ISBN 951-791-719-8.
- A:208. BIRGIT KLEYMANN: The Development of Multilateral Alliances. The Case of the Airline Industry. 2002. ISBN 951-791-720-1.
- A:209. MIKAEL EPSTEIN: Risk Management of Innovative R&D Project. Development of Analysys Model. A Systematic Approach for the Early Detection of Complex Problems (EDCP) in R&D Projects in Order to Increase Success in Enterprises. 2002. ISBN 951-791-717-9.
- A:210. SAMI KAJALO: Deregulation of Retail Hours in Finland: Historical and Empirical Perspectives. 2002. ISBN 951-791-734-1.
- A:211. TOMMI KASURINEN: Exploring Management Accounting Change in the Balanced Scorecard Context. Three Perspectives. 2003. ISBN 951-791-736-8.

- A:212. LASSE NIEMI: Essays on Audit Pricing. 2003. ISBN 951-791-751-1.
- A:213. MARKKU KAUSTIA: Essays on Investor Behavior and Psychological Reference Prices. 2003. ISBN 951-791-754-6.
- A:214. TEEMU YLIKOSKI: Access Denied: Patterns of Consumer Internet Information Search and the Effects of Internet Search Expertise. 2003. ISBN 951-791-755-4.
- A:215. PETRI HALLIKAINEN: Evaluation of Information System Investments. 2003. ISBN 951-791-758-9.
- A:216. PETRI BÖCKERMAN: Empirical Studies on Working Hours and Labour Market Flows. 2003. ISBN 951-791-760-0.
- A:217. JORMA PIETALA: Päivittäistavarakaupan dynamiikka ja ostoskäyttäytyminen Pääkaupunkiseudulla. 2003. ISBN 951-791-761-9.
- A:218. TUOMAS VÄLIMÄKI: Central Bank Tenders: Three Essays on Money Market Liquidity Auctions. 2003. ISBN 951-791-762-7.
- A:219. JUHANI LINNAINMAA: Essays on the Interface of Market Microstructure and Behavioral Finance. 2003. ISBN 951-791-783-X.
- A:220. MARKKU SALIMÄKI: Suomalaisen design-teollisuuden kansainvälinen kilpailukyky ja kansainvälistyminen. Strateginen ryhmä –tutkimus design-aloilta. 2003 ISBN 951-791-786-4.
- A:221. HANNU KAHRA: Consumption, Liquidity and Strategic Asset Allocation. 2003. ISBN 951-791-791-0.
- A:222. TONI RIIPINEN: The Interaction of Environmental and Trade Policies. 2003. ISBN 951-791-797-X.
- A:223. MIKKO SYRJÄNEN: Data Envelopment Analysis in Planning and Heterogeneous Environments. 2003. ISBN 951-791-806-2.
- A:224. ERKKI HÄMÄLÄINEN: Evolving Logistic Roles of Steel Distributors. 2003. ISBN 951-791-807-0.
- A:225 SILJA SIITONEN: Impact of Globalisation and Regionalisation Strategies on the Performance of the World's Pulp and Paper Companies. 2003. ISBN 951-791-808-9.
- A:226. EIREN TUUSJÄRVI: Multifaceted Norms in SMC Export Cooperation: A Discourse Analysis of Normative Expectations. 2003. ISBN 951-791-812-7.
- A:227. MIKA MALIRANTA: Micro Level Dynamics of Productivity Growth. An Empirical Analysis of the Great Leap in Finnish Manufacturing Productivity in 1975-2000. 2003. ISBN 951-791-815-1.
- A:228. NINA KOISO-KANTTILA: Essays on Consumers and Digital Content. 2003. ISBN 951-791-816-X.
- A:229. PETER GABRIELSSON: Globalising Internationals: Product Strategies of ICT Companies. 2004. ISBN 951-791-825-9, ISBN 951-791-826-7 (Electronic dissertation).

- A:230. SATU NURMI: Essays on Plant Size, Employment Dynamics and Survival. 2004. ISBN 951-791-829-1, ISBN 951-791-830-5 (Electronic dissertation).
- A:231. MARJA-LIISA KURONEN: Vakuutusehtotekstin uudistamisprosessi, matkalla alamaisesta asiakkaaksi. 2004. ISBN 951-791-833-X, ISBN 951-791-834-8 (Electronic dissertation).
- A:232. MIKA KUISMA: Erilaistuminen vai samanlaistuminen? Vertaileva tutkimus paperiteollisuusyhtiöiden ympäristöjohtamisesta. 2004. ISBN 951-791-835-6, ISBN 951-791-836-4 (Electronic dissertation).
- A:233. ANTON HELANDER: Customer Care in System Business. 2004. ISBN 951-791-838-0.
- A:234. MATTI KOIVU: A Stochastic Optimization Approach to Financial Decision Making. 2004. ISBN 951-791-841-0, ISBN 951-791-842-9 (Electronic dissertation).
- A:235. RISTO VAITTINEN: Trade Policies and Integration Evaluations with CGE -models. 2004. ISBN 951-791-843-7, ISBN 951-791-844-5 (Electronic dissertation).
- A:236. ANU VALTONEN: Rethinking Free Time: A Study on Boundaries, Disorders, and Symbolic Goods. 2004. ISBN 951-791-848-8, ISBN 951-791-849-6 (Electronic dissertation).
- A:237. PEKKA LAURI: Human Capital, Dynamic Inefficiency and Economic Growth. 2004. ISBN 951-791-854-2, ISBN 951-791-855-0 (Electronic dissertation).
- A:238. SAMI JÄRVINEN: Essays on Pricing Commodity Derivatives. 2004. ISBN 951-791-861-5, ISBN 951-791-862-3 (Electronic dissertation).

A-SARJA: MUITA JULKAISUJA - OTHER PUBLICATIONS

ANNE HERBERT: The Paradoxes of Action Learning: An Interpretive and Critical Inquiry into Vocational Educators' Professional Development. 2002. ISBN 951-791-684-1.

B-SARJA: TUTKIMUKSIA - RESEARCH REPORTS. ISSN 0356-889X.

- B:38. KRISTIINA KORHONEN(ed.): Current Reflections on the Pacific Rim. 2002. ISBN 951-791-661-2.
- B:39. RISTO Y. JUURMAA: Performance and International Competitiveness of Listed Metal and Telecommunication Industry Groups 1992 - 2000. Finland vs Sweden and Germany. 2002. ISBN 951-791-668-X.
- B:40. KAIJA TUOMI SINIKKA VANHALA (toim.): Yrityksen toiminta, menestyminen ja henkilöstön hyvinvointi. Seurantatutkimus metalliteollisuudessa ja vähittäiskaupan alalla. 2002. ISBN 951-791-674-4.
- B:41. ANNE ÄYVÄRI: Verkottuneen pienyrityksen markkinointikyvykkyys. 2002. ISBN 951-791-682-5.

- B:42. RIKU OKSMAN: Intohimoa ja ammattitaitoa: puheenvuoroja tuottajan työstä. 2002. ISBN 951-791-700-7.
- B:43. RISTO TAINIO KARI LILJA TIMO SANTALAINEN: Organizational Learning in the Context of Corporate Growth and Decline: A Case Study of a Major Finnish Bank. 2002. ISBN 951-791-717-1
- B:44. ELINA HENTTONEN PÄIVI ERIKSSON SUSAN MERILÄINEN: Teknologiayrittämisen sukupuoli. Naiset miesten maailmassa. 2003. ISBN 951-791-737-6.
- B:45. KIRSI KORPIAHO: "Kyllä siinä pitää elää mukana!" Kirjanpitäjien tarinoita työstä, osaamisesta ja oppimisesta työyhteisönäkökulmasta analysoituna. 2003. ISBN 951-791-742-2.
- B:46. NIILO HOME (toim.): Puheenvuoroja ECR-toiminnasta. Discussions on ECR Summaries. 2003. ISBN 951-791-749-X.
- B:47. PÄIVI KARHUNEN RIITTA KOSONEN MALLA PAAJANEN: Gateway-käsitteen elinkaari Venäjän-matkailussa. Etelä-Suomi Pietarin-matkailun väylänä. 2003. ISBN 951-791-756-2.
- B:48. ANNELI KAUPPINEN ANNE ARANTO SATU RÄMÖ (toim.): Myyttiset markkinat. 2003. ISBN 951-791-771-6.
- B:49. MIKKO SAARIKIVI SIMO RIIHONEN: Suomen puuteollisuuden kilpailukyvyn parantaminen ja kansainvälistyminen piha- ja ympäristörakentamisessa. 2003. ISBN 951-791-779-1.
- B:50. KATARIINA KEMPPAINEN ARI P.J. VEPSÄLÄINEN JUKKA KALLIO TIMO SAARINEN – MARKKU TINNILÄ: From Supply Chain to Networks: A Study of SCM Practices in Finnish Industrial Companies. 2003. ISBN 951-791-780-5.
- B:51. SAMI SARPOLA: Enterprise Resource Planning (ERP) Software Selection and Success of Acquisition Process in Wholesale Companies. 2003. ISBN 951-791-802-X.
- B:52. MATTI TUOMINEN (ed.): Essays on Capabilities Based Marketing and Competitive Superiority. Fimac II - Research: Mai Anttila, Saara Hyvönen, Kristian Möller, Arto Rajala, Matti Tuominen. 2003. ISBN 951-791-814-3.
- B:53. PÄIVI KARHUNEN RIITTA KOSONEN ANTTI LEIVONEN: Osaamisen siirtyminen Suomalais-venäläisissä tuotantoalliansseissa. Tapaustutkimuksia pietarista ja leningradin alueelta. 2003. ISBN 951-791-820-8.
- B:54. JARMO ERONEN: Kielten välinen kilpailu: Taloustieteellis-sosiolingvistinen tarkastelu. 2004. ISBN 951-791-828-3.
- B:47. PÄIVI KARHUNEN RIITTA KOSONEN MALLA PAAJANEN: Gateway-käsitteen elinkaari Venäjän-matkailussa. Etelä-Suomi Pietarin-matkailun väylänä. 2004. ISBN 951-791-846-1, korjattu painos.
- B:55. TAISTO MIETTINEN: Veron minimointi yritysjärjestelyissä. 2004. ISBN 951-791-856-9.

CKIR-SARJA: HELSINKI SCHOOL OF ECONOMICS. CENTER FOR KNOWLEDGE AND INNOVATION RESEARCH. CKIR WORKING PAPERS. ISSN 1458-5189.

- CKIR:1. SATINDER P. GILL: The Engagement Space and Parallel Coordinated Movement: Case of a Conceptual Drawing Task. 2002. ISBN 951-791-660-4.
- CKIR:2 PEKKA ISOTALUS HANNI MUUKKONEN: How Do Users of Pda's React to an Animated Human Character in Online News? 2002. ISBN 951-791-664-7.

E-SARJA: SELVITYKSIÄ - REPORTS AND CATALOGUES. ISSN 1237-5330.

- E:100. JUHA KINNUNEN: Opiskelijoiden valikoituminen pääaineisiin Helsingin kauppakorkeakoulussa. Pääainetoiveita ja niihin vaikuttavia tekijöitä kartoittava kyselytutkimus vuosina 1995-2000 opintonsa aloittaneista. 2002. ISBN 951-791-669-8.
- E:101. Research Catalogue 2000 2002. Projects and Publications. 2002. ISBN 951-791-670-1.
- E:102. DAN STEINBOCK: The U.S. CIBER Experience: The Centers for International Business Education and Research (CIBERs). 2003. ISBN 951-791-781-3.

N-SARJA: HELSINKI SCHOOL OF ECONOMICS. MIKKELI BUSINESS CAMPUS PUBLICATIONS. ISSN 1458-5383

- N:6 JUHA SIIKAVUO: Taloushallinon opas alkavalle yrittäjälle. 2002. ISBN 951-791-686-8.
- N:7. JOHANNA NISKANEN: Etelä-Savon pk-yritysten vienti Tanskaan: ulkomaankaupan erityisraportti 2001. 2002. ISBN 951-791-687-6.
- N:8. MIKKO NUMMI: Etelä-Savon pk-yritysten vienti Saksaan: ulkomaankaupan erityisraportti 2001. 2002. ISBN 951-791-688-4.
- N:9. NOORA RUOHONEN RIIKKA OLLI: Etelä-Savon pk-yritysten vienti Tsekkiin: ulkomaankaupan erityisraportti 2001. 2002. ISBN 951-791-689-2.
- N:10. ANNA HÄKKINEN ESKO LÄIKKÖ: Etelä-Savon pk-yritysten vientikohteena USA: ulkomaankaupan erityisraportti 2001. 2002. ISBN 951-791-690-6.
- N:11. JUHA SIIKAVUO: Verkko-oppimisympäristön kehittäminen. Esimerkkinä HA Boctok Venäjänkaupan erikoistumisopintojen yksi moduuli, vuosi: 2002. 2002. ISBN 951-791-695-7.
- N:12. JUHO PETTER PUHAKAINEN: German Venture Capitalists' Decision Criteria in New Venture Evaluation. 2002. ISBN 951-791-650-7.
- N:13. MILJA LEMMETYINEN: Suomalaisyrityksen etabloituminen Saksaan. Ulkomaankaupan erityisraportti 2002. 2002. ISBN 951-791-730-9.

- N:14. TAPIO PALLASVIRTA: Pk-yritysten vienti Espanjaan. Ulkomaankaupan erityisraportti 2002. 2002. ISBN 951-791-731-7.
- N:15. ELINA HAVERINEN: Etelä-Savon pk-yritysten Viron kauppa. Ulkomaankaupan erityisraportti 2003. ISBN 951-791-732-5.
- N:16. REETA RÖNKKÖ: Latinalainen Amerikka markkina-alueena Argentiina ja Brasilia. Ulkomaankaupan erityisraportti 2003. ISBN 951-791-733-3.
- N:17. JAAKKO VARVIKKO JUHA SIIKAVUO: Koulutus, oppiminen ja akateeminen yrittäjyys. 2003. ISBN 951-791-745-7.
- N:18. ANNE GUSTAFSSON-PESONEN SATU SIKANEN: Yrittäjäkoulutuksesta yrittäjäksi. 2003 ISBN 951-791-763-5.
- N:19. TOIVO KOSKI: Impact of a venture capitalists' value added on value of a venture. 2003. ISBN 951-791-764-3.
- N:20. LAURA HIRVONEN: Itävalta suomalaisyritysten markkina-alueena. 2003. ISBN 951-791-765-1.
- N:21. LAURA MALIN: Etelä-Savon pk-yritysten vienti Belgiaan. 2003. ISBN 951-791-766-X.
- N:22. JUKKA PREPULA: Ranska suomalaisten pk-yritysten vientikohteena. 2003. ISBN: 951-791-767-8.
- N:23. HENNA HUCZKOWSKI: Pk-yritysten perustaminen Puolaan. 2003. ISBN 951-791-768-6.
- N:24. HENNA KATAJA LEENA MÄÄTTÄ: Kiina suomalaisen pk-yrityksen vientikohteena. 2003. ISBN: 951-791-769-4.
- N:25. KAROLIINA IJÄS: Etelä-Savon pk-yritysten vienti Puolaan. 2003. ISBN: 951-791-770-8.
- N:26. MARJO VAHLSTEN: Matkailupalvelujen markkinoinnin kehittäminen verkkoyhteistyön avulla. 2003. ISBN: 951-791-792-9.
- N:27. TUULI SAVOLAINEN: Slovakia suomalaisten pk-yritysten markkina-alueena. 2003. ISBN: 951-791-793-7.
- N:28. HARRY MAASTOVAARA: Etelä-Savon yritysten ulkomaankauppa 2001. 2003. ISBN: 951-791-794-5.
- N:31. HANNA PERÄLÄ: Etelä-Savon pk-yritysten vienti Ruotsiin. 2003. ISBN: 951-791-799-6.
- N:34. TOIVO KOSKI ANTTI EKLÖF: Uudenmaan yrityshautomoista irtaantuneiden yritysten menestyminen, Yrittäjien näkemyksiä yrityshautomotoiminnasta sekä selvitys "yrittämisestä Työtä 2000" –projektin asiakkaiden yritystoiminnasta. 2003. ISBN 951-791-805-4.

W-SARJA: TYÖPAPEREITA - WORKING PAPERS . ISSN 1235-5674.

- W:304. PETRI BÖCKERMAN KARI HÄMÄLÄINEN MIKA MALIRANTA: Explaining Regional Job and Worker Flows. 2002. ISBN 951-791.662-0.
- W:305. PEKKA KORHONEN MIKKO SYRJÄNEN: Evaluation of Cost Efficiency in Finnish Electricity Distribution. 2002. ISBN 951-791-663-9.
- W:306. SATU NURMI: The Determinants of Plant Survival in Finnish Manufacturing. 2002. ISBN 951-791-665-5.
- W:307. JUSSI KARHUNEN: Taking Stock of Themselves. An Analysis of the Motives and the Market Reaction in Finnish Share Repurchase Programs 2002. ISBN 951-791-666-3.
- W:308. PEKKA ILMAKUNNAS HANNA PESOLA: Matching Functions and Efficiency Analysis. 2002. ISBN 951-791-671-X.
- W:309. MARKKU SÄÄKSJÄRVI: Software Application Platforms: From Product Architecture to Integrated Application Strategy. 2002. ISBN 951-791-672-8.
- W:310. MILLA HUURROS HANNU SERISTÖ: Alliancing for Mobile Commerce: Convergence of Financial Institutions and Mobile Operators. 2002. ISBN 951-791-673-6.
- W:311. ANSSI ÖÖRNI: Objectives of Search and Combination of Information Channels in Electronic Consumer Markets: An Explorative Study. 2002. ISBN 951-791-675-2.
- W:312. ANSSI ÖÖRNI: Consumer Search in Electronic Markets: Experimental Analysis of Travel Services. 2002. ISBN 951-791-676-0.
- W:313. ANSSI ÖÖRNI: Dominant Search Pattern in Electronic Markets: Simultaneous or Sequential Search. 2002. ISBN 951-791-677-9.
- W:314. ANSSI ÖÖRNI: The Amount of Search in Electronic Consumer Markets. 2002. ISBN 951-791-678-7.
- W:315. KLAUS KULTTI TUOMAS TAKALO TANJA TANAYAMA: R&d Spillovers and Information Exchange: A Case Study. 2002. ISBN 951-791-681-7.
- W:316. OLLI TAHVONEN: Timber Production v.s. Old Growth Conservation with Endogenous Prices and Forest Age Classes. 2002. ISBN 951-791-691-4.
- W:317. KLAUS KULTTI JUHA VIRRANKOSKI: Price Distribution in a Symmetric Economy. 2002. ISBN 951-791-697-3.
- W:318. KLAUS KULTTI TONI RIIPINEN: Multilateral and Bilateral Meetings with Production Heterogeneity. 2002. ISBN 951-791-698-1.
- W:319. MARKKU KAUSTIA: Psychological Reference Levels and IPO Stock Returns. 2002. ISBN 951-791-699-X.
- W:320. MERVI LINDQVIST: Possible Research Issues in Management and Control of New Economy Companies. 2002. ISBN 951-791-701-5.
- W:321. MARKO LINDROOS: Coalitions in Fisheries. 2002. ISBN 951-791-702-3.

- W:322. MIKKO SYRJÄNEN: Non-discretionary and Discretionary Factors and Scale in Data Envelopment Analysis. 2002. ISBN 951-791-705-8.
- W:323. KLAUS KULTTI HANNU VARTIAINEN: VonNeumann-Morgenstern Solution to the Cake Division Problem. 2002. ISBN 951-791-708-2.
- W:324. TOM LAHTI: A Review of the Principal-agent Theory and the Theory of Incomplete Contracts: An Examination of the Venture Capital Context. 2002. ISBN 951-791-709-0.
- W:325. KRISTIAN MÖLLER PEKKA TÖRRÖNEN: Business Suppliers' Value-Creation Potential: A Capability-based Analysis. 2002. ISBN 951-791-710-4.
- W:326. KRISTIAN MÖLLER ARTO RAJALA SENJA SVAHN: Strategic Business Nets Their Types and Management. 2002. ISBN 951-791-711-2.
- W:327. KRISTIAN MÖLLER SENJA SVAHN ARTO RAJALA: Network Management as a Set of Dynamic Capabilities. 2002. ISBN 951-791-712-0.
- W:328. PANU KALMI: Employee Ownership and Degeneration. Evidence from Estonian case studies. 2002. ISBN 951-791-713-9.
- W:329. ANNELI NORDBERG: Yrittäjyys, johtajuus ja johtaminen uuden talouden innovatiivisia haasteita. 2002. ISBN 951-791-721-X.
- W:330. LEENA LOUHIALA-SALMINEN: Communication and language use in merged corporations: Cases Stora Enso and Nordea. 2002. ISBN 951-791-722-8.
- W:331. TOMMI KASURINEN: Conceptualising the Encoding Process Related to Institutionalisation in Organisations. From Key Performance Indicator Scorecard to a Strategic Balanced Scorecard. 2002. ISBN 951-791-723-6.
- W:332. PEKKA KORHONEN HELENA TOPDAGI: Performance of the AHP in Comparison of Gains and Losses. 2002. ISBN 951-791-724-4.
- W:333. TARJA JORO PEKKA KORHONEN STANLEY ZIONTS: An Interactive Approach to Improve Estimates of Value Efficiency in Data Envelopment Analysis. 2002. ISBN 951-791-725-2.
- W:334. JUHA-PEKKA TOLVANEN JEFF GRAY MATTI ROSSI (edit.): Proceedings of the Second-Domain Specific Modeling Languages Workshop. 2002. ISBN 951-791-726-0.
- W:335. SATU NURMI: Sectoral Differences In Plant Start-up Size. 2003. ISBN 951-791-738-4.
- W:336. SATU NURMI: Plant Size, Age And Growth In Finnish Manufacturing. 2003. ISBN 951-791-739-2.
- W:337. PETRI HALLIKAINEN HANNU KIVIJÄRVI: Appraisal of Strategic it Investments: Payoffs And Tradeoffs. 2003. ISBN 951-791-740-6.
- W:338. SENJA SVAHN: Knowledge Creation in Business Networks A Dynamic-capability Perspective. 2003. ISBN 951-791-743-0.
- W:339. KRISTIAN MÖLLER SENJA SVAHN: Role of Knowledge in the Value Creation in Business Nets. 2003. ISBN 951-791-744-9.

- W:340. ELI MOEN KARI LILJA: European Works Councils in M-Real and Norske Skog: The Impact of National Traditions in Industrial Relations. 2003. ISBN 951-791-750-3.
- W:341. KJELD MÖLLER: Salatulla "arvopaperistamisella" tuhottiin yrittäjyyttä. 2003. ISBN 951 791-752-X
- W:342. ATSO ANDERSEN: Competition Between European Stock Exchanges. 2003. ISBN 951-791-753-8.
- W:343. MARKO MERISAVO: The Effects of Digital Marketing on Customer Relationships. 2003. ISBN 951-791-757-0.
- W:344. KLAUS KULTTI JUHA VIRRANKOSKI: Price Distribution in a Random Matching Model. 2003. ISBN 951-791-759-7.
- W:345. PANU KALMI: The Rise and Fall of Employee Ownership in Estonia, 1987-2001. 2003. ISBN 951-791-772-4.
- W:346. SENJA SVAHN: Managing in Networks: Case Study of Different Types of Strategic Nets. 2003. ISBN 951-791-774-0.
- W:347. KRISTIAN MÖLLER SENJA SVAHN: Crossing East-West Boundaries: Knowledge Sharing in Intercultural Business Networks. 2003. ISBN 951-791-775-9.
- W-348. KRISTIAN MÖLLER SENJA SVAHN: Managing in Emergence: Capabilities for Influencing the Birth of New Business Fields. 2003. ISBN 951-791-776-7.
- W:349. TOM RAILIO: The Taxation Consequences of Scandinavian Mutual Fund Investments and After-Tax Performance Evaluation. 2003. ISBN 951-791-777-5.
- W:350. KIRSI LAPOINTE: Subjektiivinen ura työurien tutkimuksessa ja teorioissa. 2003. ISBN 951-791-778-3.
- W:351. PANU KALMI: The Study of Co-operatives in Modern Economics: A Methodological Essay. 2003. ISBN 951-791-783-X.
- W:352. MARJA TAHVANAINEN: Short-term International Assignments: Popular Yet Largely Unknown Way Of Working Abroad. 2003. ISBN 951-791-784-8.
- W:353. MARKKU KUULA ANTOINE STAM: An Interior Point Method for Multi-party Negotiation Support. 2003. ISBN 951-791-787-2.
- W:354. JOUKO KINNUNEN: Quantification of Ordered-level Business Sentiment Survey Forecasts by Means of External Validation Data. 2003. ISBN 951-791-790-2.
- W:355. TOM RAILIO: The Nature of Disagreements and Comparability Between Finnish Accumulating Mutual Funds and Voluntary Pension Insurances. 2003. ISBN 951-791-798-8.
- W:356. JUKKA JALAVA: 'Has Our Country the Patience Needed to Become Wealthy?' Productivity in the Finnish Manufacturing Industry, 1960-2000. 2003. ISBN 951-791-803-8.
- W:357. JARI VESANEN: Breaking Down Barries for Personalization A Process View. 2003. ISBN 951-791-804-6.

- W:358. JUHA VIRRANKOSKI: Search Intensities, Returns to Scale, and Uniqueness of Unemployment Equilibrium. 2003. ISBN 951-791-809-7.
- W:359. JUHA VIRRANKOSKI: Search, Entry, and Unique Equilibrium. 2003. ISBN 951-791-810-0.
- W:360. HANNA KALLA: Exploration of the Relationship Between Knowledge Creation, Organisational Learning, and Social Capital: Role of Communication. 2003. ISBN 951-791-813-5.
- W:361. PEKKA SÄÄSKILAHTI: Strategic R&D and Network Compatibility. 2003. ISBN 951-791-817-8.
- W:362. MAIJU PERÄLÄ: Allyn Young and the Early Development Theory. 2003. ISBN 951-791-818-6.
- W:363. OSSI LINDSTRÖM ALMAS HESHMATI: Interaction of Real and Financial Flexibility: An Empirical Analysis. 2004. ISBN 951-791-827-5 (Electronic working paper).
- W:364. RAIMO VOUTILAINEN: Quantitative Methods in Economics and Management Science. 2004. ISBN 951-791-832-1 (Electronic working paper).
- W:365. MATTI KELOHARJU-SAMULI KNÜPFER-SAMI TORSTILA: Retail Incentives in Privatizations: Anti-Flipping Devices or Money Left on the Table? 2004. ISBN 951-791-839-9 (Electronic working paper).
- W:366. JARI VESANEN MIKA RAULAS: Building Bridges for Personalization A Process View. 2004. ISBN 951-791-840-2 (Electronic working paper).
- W:367. MAIJU PERÄLÄ: Resource Flow Concentration and Social Fractionalization: A Recipe for A Curse? 2004. ISBN 951-791-845-3 (Electronic working paper).
- W:368. PEKKA KORHONEN RAIMO VOUTILAINEN: Finding the Most Preferred Alliance Structure between Banks and Insurance Companies. 2004. ISBN 951-791-847-X (Electronic working paper).
- W:369. ANDRIY ANDREEV ANTTI KANTO: A Note on Calculation of CVaR for Student's Distribution. 2004. ISBN 951-791-850-X (Electronic working paper).
- W:370. ILKKA HAAPALINNA TOMI SEPPÄLÄ SARI STENFORS MIKKO SYRJÄNEN LEENA TANNER : Use of Decision Support Methods in the Strategy Process – Executive View. 2004. ISBN 951-791-853-4 (Electronic working paper).
- W:371. BERTTA SOKURA: Osaamispääoman ulottuvuudet. Arvoa luova näkökulma. 2004. ISBN 951-791-857-7 (Electronic working paper).

Y-SARJA: HELSINKI SCHOOL OF ECONOMICS. CENTRE FOR INTERNATIONAL BUSINESS RESEARCH. CIBR RESEARCH PAPERS. ISBN 1237-394X.

- Y:7. ZUHAIR AL-OBAIDI MIKA GABRIELSSON: Multiple Sales Channel Strategies in Export Marketing of High Tech SMEs. 2002. ISBN 951-791-703-1.
- Y:8. REIJO LUOSTARINEN MIKA GABRIELSSON: Globalization and Marketing Strategies of Born Globals in SMOPECs. 2004. ISBN 951-701-851-8.

Z-SARJA: HELSINKI SCHOOL OF ECONOMICS. CENTRE FOR INTERNATIONAL BUSINESS RESEARCH. CIBR WORKING PAPERS. ISSN 1235-3931.

- Z:9. V.H. MANEK KIRPALANI MIKA GABRIELSSON: Worldwide Evolution of Channels Policy. 2002. ISBN 951-791-704-X.
- Z:10. V.H. MANEK KIRPALANI MIKA GABRIELSSON: Need for International Intellectual Entrepreneurs and How Business Schools Can Help. 2004. ISBN 951-791-852-6.

Kaikkia Helsingin kauppakorkeakoulun julkaisusarjassa ilmestyneitä julkaisuja voi tilata osoitteella:

KY-Palvelu Oy Kirjakauppa Runeberginkatu 14-16 00100 Helsinki Puh. (09) 4313 8310, fax (09) 495 617 Sähköposti: kirjak@kyyppari.hkkk.fi Helsingin kauppakorkeakoulu Julkaisutoimittaja PL 1210 00101 Helsinki Puh. (09) 4313 8579, fax (09) 4313 8305 Sähköposti: julkaisu@hkkk.fi

All the publications can be ordered from

Helsinki School of Economics Publications officer P.O.Box 1210 FIN-00101 Helsinki Phone +358-9-4313 8579, fax +358-9-4313 8305 E-mail: julkaisu@hkkk.fi